



# Article Machine Learning Sensors for Diagnosis of COVID-19 Disease Using Routine Blood Values for Internet of Things Application

Andrei Velichko<sup>1,\*</sup>, Mehmet Tahir Huyut<sup>2,\*</sup>, Maksim Belyaev<sup>1</sup>, Yuriy Izotov<sup>1</sup> and Dmitry Korzun<sup>3</sup>

- <sup>1</sup> Institute of Physics and Technology, Petrozavodsk State University, 33 Lenin Ave., 185910 Petrozavodsk, Russia
- <sup>2</sup> Department of Biostatistics and Medical Informatics, Faculty of Medicine, Erzincan Binali Yıldırım University, 24000 Erzincan, Türkiye
- <sup>3</sup> Department of Computer Science, Institute of Mathematics and Information Technology, Petrozavodsk State University, 33 Lenin Ave., 185910 Petrozavodsk, Russia
- \* Correspondence: velichko@petrsu.ru (A.V.); tahir.huyut@erzincan.edu.tr (M.T.H.)

Abstract: Healthcare digitalization requires effective applications of human sensors, when various parameters of the human body are instantly monitored in everyday life due to the Internet of Things (IoT). In particular, machine learning (ML) sensors for the prompt diagnosis of COVID-19 are an important option for IoT application in healthcare and ambient assisted living (AAL). Determining a COVID-19 infected status with various diagnostic tests and imaging results is costly and timeconsuming. This study provides a fast, reliable and cost-effective alternative tool for the diagnosis of COVID-19 based on the routine blood values (RBVs) measured at admission. The dataset of the study consists of a total of 5296 patients with the same number of negative and positive COVID-19 test results and 51 routine blood values. In this study, 13 popular classifier machine learning models and the LogNNet neural network model were exanimated. The most successful classifier model in terms of time and accuracy in the detection of the disease was the histogram-based gradient boosting (HGB) (accuracy: 100%, time: 6.39 sec). The HGB classifier identified the 11 most important features (LDL, cholesterol, HDL-C, MCHC, triglyceride, amylase, UA, LDH, CK-MB, ALP and MCH) to detect the disease with 100% accuracy. In addition, the importance of single, double and triple combinations of these features in the diagnosis of the disease was discussed. We propose to use these 11 features and their binary combinations as important biomarkers for ML sensors in the diagnosis of the disease, supporting edge computing on Arduino and cloud IoT service.

**Keywords:** COVID-19; biochemical and hematological biomarkers; routine blood values; feature selection method; LogNNet neural network; machine learning sensors; Internet of Medical Things; IoT

# 1. Introduction

Identified in 2019, COVID-19 is an infectious disease caused by the novel severe acute respiratory syndrome coronavirus (SARS-CoV-2) [1,2]. Since the World Health Organization (WHO) declared the SARS-CoV-2 infection as a pandemic, the epidemic still maintains its severity to this day [3,4]. The early diagnosis of patients is extremely important to manage this unprecedented emergency [5,6]. The preferred gold standard method for detecting SARS-CoV-2 infections is the reverse polymerase chain reaction (PCR) or reverse transcriptase-PCR (RT-PCR) technique [7]. However, the execution of the test is time consuming (not less than 4–5 h under optimum conditions) and many favorable conditions must be met, such as the use of special equipment and reagents, the collection of samples and the necessity of trained personnel [8]. Machine learning (ML) and artificial intelligence (AI) models provide a powerful motivation to uncover insights from patients' data in tragic events such as the COVID-19 pandemic or in situations wherein guidelines have not yet been created [9]. ML and AI methods select the relevant biomarkers, revealing their predictive importance and consistently detecting their interactions with



Citation: Velichko, A.; Huyut, M.T.; Belyaev, M.; Izotov, Y.; Korzun, D. Machine Learning Sensors for Diagnosis of COVID-19 Disease Using Routine Blood Values for Internet of Things Application. *Sensors* 2022, 22, 7886. https:// doi.org/10.3390/s22207886

Academic Editor: Joel J. P. C. Rodrigues

Received: 15 September 2022 Accepted: 14 October 2022 Published: 17 October 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). each other. Moreover, the diagnostic performance of these methods has the ability to be improved [9–11]. AI studies for the early detection, diagnosis and prognosis of COVID-19 relied on computed tomography (CT) and RBVs. However, imaging-based solutions are costly and require specialized equipment. Machine learning (ML) and AI studies based on RBVs features are a more economical and rapid alternative method for the early detection, diagnosis and prognosis of COVID-19 [7,11,12]. Previous studies have indicated that this disease can accompany multi-organ dysfunction and cause a variety of symptoms [3,13–15]. COVID-19 can cause severe pneumonia and severe ARDS due to inflammatory cytokine storms [5,14]. The excessive and uncontrolled release of proinflammatory cytokines was considered the most important primary cause of death, similar to other infections caused by pathogenic coronaviruses [16].

The pathogen may require special attention in intensive care units (ICUs) and cause a serious respiratory disorder, in some cases leading to death [14,16]. Moreover, it is difficult to distinguish symptoms of COVID-19 from known infections in the majority of patients [14,17,18]. This predictive analytics is especially required in medical information systems (MISs) to support clinical or managerial decisions.

COVID-19 may be part of a broader spectrum of hyperinflammatory syndromes characterized by the cytokine release syndrome (CRS), such as secondary hemophagocytic lymphohistiocytosis (sHLH) [19–21]. The activation of the monocyte–macrophage system just before the disease leads to pneumonia [22,23]. During this period, changes in many routine laboratory parameters such as D-dimer and fibrinogen have been reported in COVID-19 patients [1,2,4,5,14,22,24]. High ferritin, D-dimer, lactate dehydrogenase and IL-6 levels are indicators of poor prognosis and risk of death in patients [25–27]. In addition, Winata and Kurniawan [28] reported increased D-dimer and fibrinogen degradation product (FDP) in all patients in the late stage of COVID-19. This indicates that D-Dimer and FDP levels are elevated due to increased hypoxia in severe COVID-19 conditions and are significantly associated with coagulation. Kurniawan et al. [29] reported that hyperinflammation, coagulation cascade, multi-organ failure, which play a role in the etiopathogenesis of COVID-19, and biomarkers associated with these conditions, such as CRP, D-Dimer, LDH and albumin, may be useful in predicting the outcome of COVID-19.

The previous studies detected the clinical significance of changes in the routine blood values (RBVs) in the diagnosis and prognosis of infectious diseases [1,2,4,5,30,31]. However, Jiang et al. [32], Zheng et al. [33] and Huyut [11] noted that information on early predictive RBVs should be supplemented with large samples, especially for severe and fatal cases of COVID-19.

The uncontrolled spread of the disease in pandemics distresses health systems. The early detection of patients in pandemics is an important but clinically difficult process in terms of morbidity and mortality [14,24]. The diagnosis and prognosis of COVID-19 with the use of advanced devices can provide support in improving patient comfort, health system and tackling economic inadequacies [6,11,12]. In this context, studies are carried out to diagnose and determine the severity of the disease in the early period by using ML and AI-based methods as well as RBVs data [7,11,12]. The basic element in ML approaches is to determine the feature vector with a linear classifier [30]. Since ML algorithms require a sufficiently large number of samples, the problem of dimensionality in these methods is inevitable. To minimize this problem, the dataset should be reduced by finding a less dimensional attribute matrix. The dimensionality problem can be minimized by discarding irrelevant features with the feature selection procedure [30,31].

Feature selection methods can be summarized under three main headings: embedded methods, filters and wrappers (backward elimination, forward selection, recursive feature elimination) [30,31]. Feature selection in embedded methods is part of the training process and, therefore, this method lies between filters and wrappers. In the embedded methods, the determination of the best subset of features is performed during the training of the classifier (for example, when optimizing weights in a neural network). In terms of computational cost, embedded methods are more economical than wrappers [30]. Although we can find many case studies for all three feature selection methods, most feature selection methods are filters [30]. The existence of a large number of available feature selection methods complicates the selection of the best method for a particular problem [31]. The popular feature selection methods include correlation-based feature selection (CFS) [34], consistency-based filtering [35], INTERACT [36], information gain [37], ReliefF [38], recursive feature elimination for support vector machines (SVM-RFE) [39], Lasso editing [40] and minimum redundancy maximum relevance (mRMR) algorithm (developed specifically for dealing with microarray data) [30].

We examined the SARS-CoV-2-RBV1 database using the LogNNet neural network [12]. LogNNet can be defined as a feed forward network that increases the classification accuracy by chaotic mapping that fills a reservoir matrix. It is important to optimize the chaotic map parameters in data analysis by applying the LogNNet neural network. In addition, by taking advantage of chaotic mapping, it is possible to significantly reduce the RAM usage by a neural network. These results show that LogNNet can be used effectively in Internet of Things (IoT) mobile devices.

The main point for many digital health solutions during the pandemic process is the production of effective, fast and inexpensive alternative methods for the early diagnosis and treatment of COVID-19 patients. However, even the most knowledgeable and experienced physicians can interpret little of the information contained in routine blood laboratory results, and it is extremely difficult to determine the severity of COVID-19 patients based on RBVs findings alone [41]. In this context, ML classification models run with RBV-based data to determine the preliminary diagnosis of COVID-19 can be an effective tool in clinical decision support systems with an accuracy of over 95%. In this study, 13 ML models and LogNNet neural networks were applied in the diagnosis of suspected cases with an alternative device, based on LogNNet and Andrunio solutions, as only RBV-based, and the most important features were determined. We made a clinical interpretation of the relationship between these features and their various combinations with the disease. We achieved the performance of all models in detecting the diagnosis of the disease and reached up to 99.8% accuracy. ML sensors (Sensors 1.0 type) for the diagnosis of the COVID-19 disease have been successfully tested in the IoT environment, and the diagnosis of the disease has been implemented in offline and online modes. In offline mode, ML sensors were run on an Arduino board with a LogNNet neural network with a total RAM consumption of ~4 kB. Obtaining the findings in this study over a large sample is an important advantage in terms of the validity of the study. We believe that this study will help to identify suspected patients with a high probability of being infected with COVID-19 at the time of admission to the hospital with a fast and economical method, which will make important contributions to the detection of the disease before it progresses.

The paper has the following structure. Section 2 present the related studies, Section 3 describes the data collection procedure, correlation analysis of features, machine learning methods and the implementation of LogNNet on an Arduino board. Section 4 presents the results from the correlation analysis of dataset, classification results, one, double, triple and 11 feature combinations in the detection of sick and healthy individuals, and the ML sensor concept for IoT. Section 5 discusses the results and compares them with known developments. Section 6 presents the limitations of the study. In conclusion, a general description of the study and its scientific significance are given.

### 2. Related Studies

The prompt diagnosis of COVID-19 seems to be a promising advancement for applying at-home health care and AAL [42]. The digitalization of healthcare calls for effective applications of human body sensors [43] and human sensing [44], including ML sensors, to continuously monitor various parameters of the human body in everyday life with the help of the IoT [45]. Everyday human body sensors testify to the growing number of applications in IoT-enabled ambient intelligence (AmI) systems [46]. The paradigms of ML sensors [47] and artificial intelligence (AI) sensors [48,49] are similar in meaning. The ML

sensor paradigm was further developed by Warden et al. [47] and Matthew Stewart [50], wherein the authors introduced the terms Sensors 1.0 and Sensors 2.0 devices. Sensors 2.0 devices are a combination of both a physical sensor and a machine learning module in one package. Sensors 2.0 devices process data internally, ensuring data security, while in Sensors 1.0 devices, these modules are physically separated. In addition, the authors proposed the concept of creating a datasheet of ML sensors. Therefore, the development of technology for creating ML sensors for the diagnosis of the COVID-19 disease is an urgent problem.

In previous studies, the RBVs of people who lived and died from COVID-19, or patients with COVID-19 and healthy individuals, were statistically compared [1,3,14,22–24,26,51]. In addition, differences in many RBVs characteristics are known between mild and severe COVID-19 patient groups according to statistical methods. However, this study demonstrates that ML models using only one or two features can detect COVID-19 patients from a large group of patients with high accuracy. Therefore, this study will be an alternative approach with extremely high sensitivity for the diagnosis of COVID-19. ML algorithms allow for an easy interpretation of complex association structures in data by simultaneously evaluating the cumulative effects of numerous biomarkers to discover higher-order interactions [4,6,9]. With this benefit, the strengths of using ML in clinical medicine are considered as an opportunity. Although various clinical studies [7,52,53] have highlighted how blood test-based diagnosis can provide an effective and low-cost alternative for the early detection of COVID-19 cases, relatively few ML models have been applied to blood parameters [7,54].

An evaluation of lung CT images to predict lung cancer using deep learning with an improved abundant clustering technique and instant trained neural networks approach was performed in [55]. The authors achieved an accuracy of up to 98.42% in cancer diagnosis with a minimum classification error of 0.038. Cui et al. [56] examined the distribution of pixels in the images with the fuzzy Markov random field segmentation approach using positron emission tomography (PET) and computed tomography (CT) images of the affected area associated with lung tumor. The developed method provided an accuracy of 0.85 in recognizing the lung tumor region. Tomita et al. [57] ran a logistic regression (LR) analysis, support vector machine (SVM) and deep neural network (DNN) models with biochemical findings, lung function tests and bronchial challenge test features to predict the initial diagnosis of adult asthma. In the pre-diagnosis of adult asthma, the DNN model showed 0.98%-ACC, the SVM model 0.82%-ACC and the LR model 0.94%-ACC. Ryu et al. [58] used various ML models and a deep neural network model for the pre-diagnosis of diabetes mellitus using various physical and routine blood values features. The deep neural network has been the most successful model with a value of 0.80-AUC in diabetes mellitus. Kolachalama et al. [59] used a six-convolutional deep learning architecture (CNN) with histological images, biopsy results and some clinical phenotypes to classify kidney disease severity. The CNN model was found to be more successful with AUC values of 0.878, 0.875 and 0.904, respectively, than the pathologist-predicted fibrosis score (0.811, 0.800 and 0.786 AUC, respectively) for assessing 1-, 3- and 5-year renal survival. In a study conducted to identify patients at risk of early diagnosis of fatty liver disease, Wu et al. [60] used an artificial neural network model with three ML models. The most successful model in the diagnosis of risky patients was the random forest with 87.48-ACC and 0.92-AUC values. Oguntimilehin et al. [61] used an ML technique on a set of labeled typhoid fever contingent variables for the diagnosis of typhoid fever and to establish explicable rules. The labeled database is divided into five different levels of typhoid fever severity, with classification accuracies on both the training set and the test set of 95% and 96%, respectively. The application was implemented using Visual Basic as the front-end and MySQL as the back-end. Kouchaki et al. [62] used various ML methods to predict the resistance to MTB in Mycobacterium tuberculosis (MTB) patients given a specific drug in a timely manner and to identify resistance markers. Compared to the traditional molecular diagnostic test, the AUC values of the best ML classifiers were higher for all

drugs. Logistic regression and gradient tree reinforcement methods performed better than other techniques. Taylor et al. [63] ran six machine learning algorithms with 10 features consisting of patient demographics, RBVs results and drug information for the diagnosis of and treatment decisions for urinary tract infection. The best performing model, XGBoost, diagnosed the presence of a urinary tract infection with a high AUC value (0.826–0.904 confidence interval).

Yang et al. [64] ran four ML models on 3356 patients (42% COVID-19 positive) using 27 features covering both blood count and biochemical parameters. A gradient boosted decision tree model was the most successful model in the diagnosis of the disease with a value of 0.85-AUC. Booth et al. [9] operated 26 RBVs data elements with a support vector machine to detect COVID-19 patients at high mortality risk and determined prognostic biomarkers with a value of 0.93-AUC. Huyut [11] classified severely and mildly infected patients from a large population of COVID-19 patients using 12 supervised ML models and 28 routine blood values. The models with the highest AUC for identifying mildly infected patients were found to be local weighted learning at 0.95%, Kstar at 0.91%, Naive bayes at 0.85% and K nearest neighbor at 0.75%. Brinati et al. [65] ran 13 RBVs with various ML methods to detect COVID-19 patients (102 negative and 177 COVID-19 positive). They noted that the models with the highest accuracy in the diagnosis of the disease were random forest (82%) and logistic regression (78%). Huyut and Velichko [12] determined the diagnosis and prognosis of the COVID-19 disease by running the LogNNet neural network model on 51 RBVs features. The model achieved an accuracy of 99% in the diagnosis of the disease and 84% in its prognosis. Zhang et al. [66] used a variety of demographic indicators and 21 RBVs using a Lasso-based neural network model to detect predictors of mortality from COVID-19. The success of the model in determining the clinical status of the patients was 98%-AUC. Alle et al. [67] applied the XGboost and logistic regression model on a dataset of various clinical and laboratory tests to predict COVID-19 mortality and found accuracy rates of 83% and 92%, respectively. Gao et al. [68] applied an ensemble model derived from support vector machine (SVM), gradient augmented decision tree (GBDT) and neural network (NN) algorithms using 28 immune/inflammatory features to detect COVID-19. The developed model reached 0.99 AUC in detecting infected patients. Vaishnav et al. [69] used various machine learning models to predict mortality from COVID-19, and the decision tree regression model produced a 70% accuracy and the random forest regression model a 76% accuracy. Huyut and İlkbahar [5] used various biomarkers with the CHAID decision tree to detect the diagnosis and prognosis of COVID-19. The model produced an 81.6% accuracy in recognizing the disease and a 93.5% accuracy in determining the prognosis of the disease. Huyut and Üstündağ [6] used 23 blood gas parameters with the CHAID decision tree to predict the diagnosis and prognosis of COVID-19. The model produced a 68.2% accuracy in recognizing the disease and a 65.0% accuracy in determining the prognosis of the disease. Kukar et al. [70] constructed a machine learning model based on 35 RBVs to diagnose 5333 negative and 160 positive COVID-19 patients with various bacterial and viral infections. The model showed an 81.9% sensitivity and a 97.9% specificity in detecting patients. Mei et al. [71] developed a model combining CNN and multilayer sensor to detect COVID-19 using computed tomography (CT), various clinical information elements and some RBVs data. The model reached an 84% sensitivity and an 83% specificity in recognizing the disease.

AI studies on the risk of poor outcome for COVID-19 patients need further validation with larger samples [11,25,72]. Furthermore, previous AI studies using RBVs for COVID-19 diagnosis and prognosis which covered the early stages of the outbreak included less blood values and reported poorer performance. Therefore, to detect the disease in the later stages of the epidemic, it is necessary to study ML models on a larger sample, which can achieve higher accuracy and use most RBVs.

## 3. Data and Methods

The data used in this study were collected retrospectively from the information system of Erzincan Binali Yıldırım University Mengücek Gazi Training and Research Hospital (EBYU-MG) between April and December 2021. The data used in this study are shared as open access under the name of "SARS-CoV-2-RBV1" in [12].

During the dates covered by this study, a diagnosis of SARS-CoV-2 was made by real-time reverse transcriptase polymerase chain reaction (RT-PCR) on nasopharyngeal or oropharyngeal swabs at the EBYU-MG hospital. RBVs results at first admission were recorded to prevent various complications.

## 3.1. Characteristic of Participants, Workflow and Datasets

Between the specified dates, the digital system of our hospital was scanned and patients diagnosed with COVID-19 (n = 2648) were selected from a large patient population (a dataset of approximately 80 thousand patients was scanned). The routine laboratory information of these patients was examined. The parameters (features) that were measured from at least 80% of the patients were used. Missing data were completed with the mean of the distribution and normalized. A total of 51 routine blood values calibrated from approximately 70 parameters were recorded. In addition, a group (control group) with the same number of negative COVID-19 tests (n = 2648) was identified and 51 characteristics of these individuals were recorded. Our control group arrived at the hospital only with the suspicion of COVID-19. Chronic disease information of the patients could not be reached. Only data of individuals over the age of 18 were recorded.

These two datasets were combined and named "SARS-CoV-2-RBV1" dataset. The SARS-CoV-2-RBV1 dataset includes immunological, hematological and biochemical RBVs parameters and consists of 51 features (Table 1). In the SARS-CoV-2-RBV1 dataset, positive COVID-19 test results were coded as 1 and negative test results as 0 (COVID-19 = 1, non-COVID-19 = 0).

№	Feature	ւթ	Feature	№	Feature	№	Feature	№	Feature
1	CRP	12	NEU	23	MPV	34	GGT	45	Sodium
2	D-Dimer	13	PLT	24	PDW	35	Glucose	46	T-Bil
3	Ferritin	14	WBC	25	RBC	36	HDL-C	47	TP
4	Fibrinogen	15	BASO	26	RDW	37	Calcium	48	Triglyceride
5	INR	16	EOS	27	ALT	38	Chlorine	49	eGFR
6	PT	17	HCT	28	AST	39	Cholesterol	50	Urea
7	РСТ	18	HGB	29	Albumin	40	Creatinine	51	UA
8	ESR	19	MCH	30	ALP	41	СК		
9	Troponin	20	MCHC	31	Amylase	42	LDH		
10	aPTT	21	MCV	32	CK-MB	43	LDL		
11	LYM	22	MONO	33	D-Bil	44	Potassium		

Table 1. Feature numbering for SARS-CoV-2-RBV1 dataset [12].

CRP: C-reactive protein; INR: international normalized ratio; PT: prothrombin time; PCT: procalcitonin; ESR: erythrocyte sedimentation rate; aPTT: activated partial prothrombin time; LYM: lymphocyte count; NEU: neutrophil count; PLT: platelet count; WBC: white blood cell count; BASO: basophil count; EOS: eosinophil count; HCT: hematocrit; HGB: hemoglobin; MCH: mean corpuscular hemoglobin; MCHC: mean corpuscular hemoglobin concentration; MCV: mean corpuscular volume; MONO: monocyte count; MPV: mean platelet volume; PDW: platelet distribution width; RBC: red blood cells; RDW: red cell distribution width; ALT: alanine aminotransaminase; AST: aspartate aminotransferase; ALP: alkaline phosphatase; CK-MB: creatine kinase myocardial band; D-Bil: direct bilirubin; GGT: gamma-glutamyl transferase; HDL-C: high-density lipoprotein cholesterol; CK: creatine kinase; LDH: lactate dehydrogenase; LDL: low-density lipoprotein; T-Bil: total bilirubin; TP: total protein; eGFR: estimating glomerular filtration rate; UA: uric acid. The features in this dataset are calibrated and contain almost all of the routine blood values that are the subject of studies on COVID-19 in the literature. Therefore, we believe that the bias of our study using this dataset was minimized in comparison to the literature. In addition, the use of our dataset, which we share as open access, is important in terms of showing the reproducibility and auditability of the results.

# 3.2. Correlation Analysis of Features

To determine the level of correlation between diagnosis and biochemical blood parameters, the original dataset was analyzed using the point-biserial correlation test [73]. Pearson correlation coefficient was calculated for each feature–feature pair, and a correlation matrix was compiled. The correlation matrix makes it possible to judge the strength and structure (positive or negative) of the linear relationship between diagnosis–feature and feature–feature pairs. The correlation matrix was created using the pandas software package [74].

#### 3.3. Machine Learning Methods, Hyperparameters, Accuracy Estimation

Machine learning algorithms can be applied to a wide range of problems such as classification, clustering, regression analysis, time series forecasting, etc. [75]. The SARS-CoV-2-RBV1 dataset under study has an output parameter divided into two classes (positive or negative diagnosis for COVID-19), so the task of the machine learning algorithm is reduced to binary classification based on 51 features. This study compared the accuracy of the most popular binary classification algorithms: multinomial naive Bayes (MNB), Gaussian naive Bayes (GNB), Bernoulli naive Bayes (BNB), linear discriminant analysis (LDA), K-nearest neighbors (KNN), support vector machine classifier with linear kernel (LSVM), support vector machine classifier with non-linear kernel (NLSVM), passive-aggressive (PA), multilayer perceptron (MLP), decision tree (DT), extra trees (ETs) classifier, random forest (RF), histogram-based gradient boosting (HGB).

Each classifier model has hyperparameters, for which optimization is necessary to obtain the most accurate models. For optimization, the software package "auto-sklearn" [76] was used.

Before training the models, the initial data were subjected to preprocessing, which makes it possible to speed up the training of the models and improve the accuracy of the classification. Preprocessing includes two stages: (1) normalization of numerical values of the input data, (2) generation of additional features. Normalization is a procedure consisting of bringing numerical data to a single format, which has the following options: quantile transformer (QT)—transforms feature values so that they correspond to a uniform or normal distribution; robust scaler (RS)—subtracts the median values for each feature and scales according to the interquartile range; MinMax (MM)—scales feature values so that they are all in the range from the minimum to the maximum value. The procedure for generating additional features transforms the original set of features into a set of features with a different dimension. This helps to select the most important features, compose additional features from them or present the input data in a special format for the ML algorithm. The following methods for generating additional features were used: polynomial (PN) creates features that are polynomial combinations of the original features; random trees embedding (RTE)—creates a multidimensional sparse feature representation, in which the data in each new feature are represented by binary values; extra trees preprocessor (ETP)—selects a part of the most important features that are evaluated using the extra trees algorithm; linear SVM preprocessor (LSVMP)—selects some of the most important features that are evaluated using the support vector machine algorithm; independent component analysis (ICA)—selects a set of statistically independent features from the entire original set; Nystroem sampler (NS)—transforms a set of initial features using a low-rank matrix approximation by the Nystroem method.

The accuracy of models  $A_{\rm NF}$  was assessed by the K-fold cross-validation method (K = 5) encapsulated in software packages, wherein the designation  $A_{\rm NF}$  refers to the

classification accuracy when using *NF* features. K-fold cross-validation method splits the original dataset into *K* parts and sequentially trains the model. One of the *K* parts of the dataset is used as a test sample, and the other parts as a training sample. Then, the obtained values of the classification accuracy on the test samples are averaged. The division of the base into parts is performed using stratification. Such approach makes it possible to reliably estimate the accuracy of models.

In this study, we used a less common ML algorithm based on the LogNNet neural network. The LogNNet 51:50:20:2 configuration was used, and a detailed configuration description is given in [75]. The LogNNet architecture is IoT-oriented and can run on devices with low computing resources (Section 3.4).

Each algorithm was given the same amount of time (1 h) to optimize the hyperparameters. A computer with an AMD Ryzen 9 3950X processor and 64 GB DDR-4 RAM was used to train the models.

#### 3.4. Implementing LogNNet on an Arduino Board

The Arduino Nano 33 IoT board was chosen as a prototype IoT edge device with limited computing resources. It is based on a 32-bit Microchip ATSAMD21G18 microcontroller with an ARM Cortex-M0+ computing core, a clock frequency of 48 MHz, 256 KB of flash memory and 32 KB of RAM. The neural network LogNNet 51:50:20:2 from [12] was programmed on the Arduino board and tested. Arduino Nano 33 IoT test circuit, LogNNet architecture and board are presented in Figure 1.



Figure 1. Arduino Nano 33 IoT test circuit and LogNNet architecture.

3.4.1. LogNNet Program for Arduino Board

LogNNet transforms the input feature vector d into a normalized vector Y, which is multiplied with the reservoir matrix W filled with a chaotic mapping. We used the mapping congruent generator (1) with the parameters indicated in Table 2 and the data [12]. Then, the transformed vector passes the output classifier (two-layer feed-forward neural network with two hidden layers).

Table 2. Chaotic map equation and list of parameters with limits.

Chaotic Map	List of Parameters	Equation	
Congruent generator	K = 93 D = 68 L = 9276 C = 73	$\begin{cases} x_{n+1} = (D - K \cdot x_n) \mod L \\ x_1 = C \end{cases}$	(1)

Let us denote the matrix of weight coefficients between the layers  $S_h$  and  $S_{h2}$  as  $W_1$ , and the matrix between the layers  $S_{h2}$  and  $S_{out}$  as  $W_2$ . At the output, there are two neurons

for two classes (COVID-19 and non-COVID-19). Matrices of weight coefficients and values of normalization coefficients were calculated on a computer with high performance and saved in a separate library file. In addition, the library file (supplementary materials) contains the values of *K*, *D*, *L* and *C* required for calculating the *W* matrix, as well as data on configuration of the LogNNet 51:50:20:2 neural network.

When LogNNet is running, the values of the elements of the matrix W (2550 values) are sequentially calculated using the congruent generator method (1) each time a feature vector is input. This approach does not store the matrix W in the RAM memory of the controller, and it leads to memory saving; however, it slows down the calculations of the neural network.

The Arduino IDE development environment was used to implement the algorithm. The library file with the matrices  $W_1$  and  $W_2$  and other coefficients necessary for the operation of the neural network were loaded at the beginning of the program. The complete code of the program is presented in Appendix A, Algorithm A1. The algorithm is divided into functions and procedures:

- Function "Fun\_activ"—activation function, lines 10–12;
- Procedure "Reservoir"—calculation of coefficients of reservoir matrix W by congruent generator formula, multiplication of arrays and calculation of neurons in layer Sh, lines 14–28;
- Procedure "Hidden\_Layer"—calculation of neurons in the hidden layer Sh2, lines 30–39;
- Function "Output\_Layer\_Layer"—calculation of the output layer Sout, lines 41–54;
- The "void loop" block is an executable loop, lines 61–77;
- "void setup" block—initialization block, lines 61–77.

The scaling factor "scale\_factor = 1000" makes it possible to convert data from a floating point type to an integer (and vice versa), by multiplying (dividing) by a factor and rounding. In the Arduino, a float variable takes 4 bytes of RAM, and an integer variable takes 2 bytes of RAM. Therefore, storing matrices  $W_1$ ,  $W_2$  and other data in integer format are more efficient, and during library initialization, the data takes 2 times less RAM memory.

#### 3.4.2. Test Scheme

Neural network testing is the serial sending of SARS-CoV-2-RBV1 data to the Arduino board and counting the correct network responses. The data is generated on an external computer (Figure 1). For sending data, a protocol was implemented that separates the elements of the feature vector *Y* using the symbol "T" to avoid data gluing. At the end of the vector *Y*, special characters "FN" are placed, indicating the end of the data transfer. On the Arduino side, a protocol is implemented that recognizes the input data. In the "void loop" block, a loop is organized to check the availability of data in the serial port buffer using the Serial.Available function. This function returns "True" as soon as the Arduino receives data.

## 4. Results

# 4.1. Correlation Analysis of Dataset SARS-CoV-2-RBV1

Figure 2 presents the results of the correlation analysis of the diagnosis–feature and feature–feature pairs in the form of a "heatmap" over the entire volume of the SARS-CoV-2-RBV1 database. The first column of Table 3 shows the features most highly associated with the survivors and non-survivors of COVID-19 with a point-biserial correlation ( $r_{\rm pb}$ ) coefficient exceeding 0.5. Here, the negative or positive result of the point-biserial correlation coefficient provides information about the direction of the relationship between the diagnosis of the disease and the quantitative characteristics. As seen in the first column of Table 3, the features most associated with disease diagnosis are MCHC, HDL-C, cholesterol and LDH. The second column of Table 3 shows the accuracy of the cut-off values calculated by the threshold approach [12] for each trait in classifying COVID-19 patients. The features

presented in the second column are the predictors that classify patients with the highest accuracy. For comparison, the results of the threshold classification  $A_{th}$  from [12] are presented, and features with  $A_{th} \ge 70\%$  are shown. The threshold classification method and the point-biserial correlation method give an intersecting set of features, but the threshold classification provides more diagnosis-related features. While the point-biserial correlation coefficient reveals the level of association between living and deceased COVID-19 patient traits, the diagnosis has only two values (1 and 0). However, when separating these two classes, the threshold method considers all the data of the relevant feature, and it has high sensitivity.

**Table 3.** Features most strongly correlated with the diagnosis according to the point-biserial correlation coefficient and the threshold correlation method.

Feature (Point-Biserial Correlation Coefficient (r <sub>pb</sub> ))	Feature (Threshold Accuracy of Classification A <sub>th</sub> from [12])
MCHC (0.8)	MCHC (94.35%)
HDL-C (-0.77)	HDL-C (94.73%)
Cholesterol $(-0.71)$	Cholesterol (94.47%)
LDL (-0.68)	LDL (96.47%)
	Triglyceride (90.96%)
	Amylase (85.1%)
	UA (81.12%)
	TP (79.68%)
	СК-МВ (78.91%)
	LDH (74.98%)
	Albumin (72.91%)



**Figure 2.** Correlation of the SARS-CoV-2-RBV1 dataset for diagnosis–feature (point-biserial correlation) and feature–feature pairs (Pearson coefficient).

An analysis of the correlation of features among themselves (Figure 2) reveals several features that are linearly dependent on each other. The most strongly correlated pairs with the Pearson coefficient exceeding 0.6 modulo are shown in Table 4. The same table presents Pearson's coefficients separately for a variety of COVID-19 positive and negative participants. Full heatmaps by class (COVID-19, non-COVID-19) are shown in Figures 3 and 4.

**Table 4.** The features most strongly correlated with each other by the Pearson coefficient for the entire database and separately for classes (positive or negative COVID-19 status).

Pair Feature–Feature	Pearson's Coefficient for COVID-19 Diagnosis	Pearson's Coefficient for Positive COVID-19	Pearson's Coefficient for Negative COVID-19
		Type High–High	
HCT-HGB	0.96	0.95 (High)	0.97(High)
MPV-PDW	0.93	0.94	0.92
HCT-RBC	0.87	0.88	0.87
MCH-MCV	0.84	0.84	0.84
HGB-RBC	0.83	0.83	0.83
NEU–WBC	0.74	0.71	0.81
Albumin–TP	0.64	0.67	0.5
MCH-MCHC	0.53	0.62	0.99
MCH-RDW	-0.55	-0.61	-0.51
		Type High–Low	
Fibrinogen–LYM	-0.77	-0.78 (High)	-0.01 (Low)
Cholesterol-LDL	0.65	0.59	0.012
Cholesterol-HDL-C	0.64	0.39	-0.024
Chlorine-Sodium	0.18	0.63	-0.025
		Type Low–High	
MCHC-MCV	0.41	0.09 1(Low)	0.84 (High)
ALT-AST	0.6	0.48	0.76
eGFR–Urea	-0.55	-0.49	-0.63
INR-PT	0.12	0.075	1
D-Bil–T-Bil	0.6	0.33	0.91
HDL-C-LDL	0.63	0.19	0.3

Three main types of pair correlations can be distinguished. The High–High type is the pairs of features for which the correlation has a high value does not depend on the presence or absence of COVID-19 disease. The High–Low type is the pairs of features that are highly correlated only in sick patients. The Low–High type is the pairs of features that are highly correlated only in healthy patients. In general, the features are more correlated in patients with COVID-19 (Figure 3). From a medical point of view, pair correlation will be reviewed in the Discussion Section.



**Figure 3.** Pearson correlation analysis results for positive diagnoses for COVID-19 from the SARS-CoV-2-RBV1 dataset.



**Figure 4.** Pearson correlation analysis results for negative diagnoses for COVID-19 from the SARS-CoV-2-RBV1 dataset.

# 4.2. Classification Results for Dataset SARS-CoV-2-RBV1

Table 5 presents the results of the machine learning algorithms optimized to obtain the maximum classification values using 51 features. The results are sorted in descending order of algorithm efficiency. For each algorithm, the average training and inference time of the model and methods for preprocessing of the input data are given.

Classification Algorithm	Average Model Accuracy A <sub>51</sub> ,%	Average Learning Time, s	Average Inference Time, μs	Normalization Method	Methods for Generating Additional Features
Histogram-based Gradient Boosting	100	6.39	11.6	-	-
Random Forest	99.943	13.15	21.9	QT	-
K-nearest neighbors	99.924	3.17	22.1	QT	ETP
Extra Trees classifier	99.905	18.73	24.5	RS	-
Multilayer Perceptron	99.886	3.99	2.2	RS	LSVMP
Multinomial Naive Bayes	99.792	2.48	11.7	QT	RTE
Linear Discriminant Analysis	99.773	9.15	7.5	QT	PN
Support Vector Machine with non-linear kernel	99.754	222.41	43.2	QT	NS
Decision Tree	99.660	1.46	1.2	RS	LSVMP
Passive-Aggressive	99.641	2.91	13.1	QT	RTE
Bernoulli Naive Bayes	99.622	2.59	11.3	QT	RTE
Support Vector Machine with linear kernel	99.584	5.21	1242	MM	PN
Gaussian Naive Bayes	98.565	1.68	3.5	QT	ICA
LogNNet [12]	99.509	100	3	-	-

**Table 5.** Results of assessing the classification accuracy of machine learning models for the SARS-CoV-2-RBV1 dataset.

The accuracy of the algorithms  $A_{51}$  ranged from 98.56% to 100%, indicating that all models were good at identifying the association of features with the diagnosis of COVID-19. The most efficient model is based on the histogram-based gradient boosting classifier with a 100% accuracy.

Figure 5 presents the learning curves for the HGB model using all the features from the dataset. The red curve (training accuracy) shows the training ability of the model, and the green curve (cross-validation accuracy) shows the generalization ability of the model depending on the number of training examples. Each point on the graph was obtained using five different splits into a test (20%) and training (80%) subsets.



**Figure 5.** Learning curves for the histogram-based gradient boosting classifier model using 51 features from the SARS-CoV-2-RBV1 dataset.

The red curve represents the accuracy of the model on the training samples. The model has sufficient complexity to recognize all training samples with a 100% accuracy. The green curve represents the accuracy of the model on the test subset, the samples of which were not involved in model training. With an increase in training samples, the cross-validation accuracy of the model grows. The curves converge with each other and completely coincide when the number of training samples is more than 2500. The dots on the graph represent the average accuracy using five different splits, and the shaded areas represent the standard deviation.

Unlike other models, HGB does not require data preprocessing. The training time of the HGB model is about 6 s, which makes it possible to effectively use it to enumerate input features when searching for optimal combinations. The LogNNet model was used to implement the classification on the Arduino board, so its algorithm has a compact form suitable for IoT devices.

The HGB model was used to study the most significant combinations of the first, second and third features.

# 4.2.1. Investigation of the Effectiveness of the HGB Model Operating on One Feature

Table 6 presents the classification result of the SARS-CoV-2-RBV1 dataset for the HGB model using a single input feature. The features are sorted in descending order of  $A_1$  classification accuracy. The most effective features are the first six features: LDL ( $N^{\circ}$  43), cholesterol ( $N^{\circ}$  39), HDL-C ( $N^{\circ}$  36), MCHC ( $N^{\circ}$  20), triglyceride ( $N^{\circ}$  48) and amylase ( $N^{\circ}$  31). The same features are dominant in assessing the correlation between the sign and the diagnosis from Table 3.

**Table 6.** Classification efficiency of SARS-CoV-2-RBV1 datasets using the single feature for the Histogram-based Gradient Boosting classifier.

N	Feature	<i>A</i> <sub>1</sub> ,%	N	Feature	<i>A</i> <sub>1</sub> ,%	.№	Feature	<i>A</i> <sub>1</sub> ,%	₽	Feature	<i>A</i> <sub>1</sub> ,%
43	LDL	96.84	4	Fibrinogen	76.03	50	Urea	68.10	21	MCV	56.43
39	Cholesterol	95.07	29	Albumin	75.3	7	PCT	63.25	22	MONO	56.26
36	HDL-C	94.99	44	Potassium	75.22	27	ALT	62.33	5	INR	56.19
20	MCHC	94.35	3	Ferritin	74.45	35	Glucose	62.17	6	PT	56.04
48	Triglyceride	93.76	38	Chlorine	73.18	49	eGFR	62.04	17	HCT	55.75
31	Amylase	90.01	46	T-Bil	72.77	14	WBC	61.91	26	RDW	55.62
51	UA	87.91	34	GGT	72.62	16	EOS	61.40	9	Troponin	54.07
42	LDH	85.76	41	СК	70.97	13	PLT	61.25	18	HGB	53.94
47	TP	80.41	2	D- Dimer	70.46	28	AST	60.55	25	RBC	53.43
37	Calcium	80.40	33	D-Bil	70.37	8	ESR	59.12	23	MPV	53.13
32	CK-MB	79.73	11	LYM	69.90	15	BASO	58.72	24	PDW	53.09
1	CRP	77.81	45	Sodium	69.35	12	NEU	57.51	19	MCH	52.13
30	ALP	77,71	40	Creatinine	69,24	10	aPTT	56.53			

4.2.2. Investigation of the Effectiveness of the HGB Model Operating on Two Features

Table 7 presents the classification result of the SARS-CoV-2-RBV1 dataset for the HGB model using two input features. The pairs of features are sorted in descending order of classification accuracy  $A_2$ .

N≞	First Feature	Second Feature	Average Accuracy $A_2,\%$
20-19	MCHC	MCH	99.81
43-32	LDL	CK-MB	99.62
36-32	HDL-C	CK-MB	99.49
48-32	Triglyceride	CK-MB	99.45
43-39	LDL	Cholesterol	99.43
43-20	LDL	MCHC	99.22
39-36	Cholesterol	HDL-C	99.18
39-48	Cholesterol	Triglyceride	99.11
43-42	LDL	LDH	99.05
43-31	LDL	Amylase	99.03
36-20	HDL-C	MCHC	98.98
43-51	LDL	UA	98.86
36-31	HDL-C	Amylase	98.81
39-20	Cholesterol	MCHC	98.73
20-48	MCHC	Triglyceride	98.65
39-38	Cholesterol	Chlorine	98.62
43-38	LDL	Chlorine	98.43
20-31	MCHC	Amylase	98.28
36-42	HDL-C	LDH	98.16
48-42	Triglyceride	LDH	98.14

 

 Table 7. Classification efficiency of SARS-CoV-2-RBV1 dataset using 2 features for the Histogrambased Gradient Boosting classifier.

The resulting pairs contain the most effective features: LDL ( $N_{\text{P}}$  43), cholesterol ( $N_{\text{P}}$  39), HDL-C ( $N_{\text{P}}$  36), MCHC ( $N_{\text{P}}$  20), triglyceride ( $N_{\text{P}}$  48) and amylase ( $N_{\text{P}}$  31), which have the best  $A_1$  score (Table 7). The best result ( $A_2$  = 99.81) was obtained for the MCHC–MCH feature pair. At the same time, the pair contains the MCH ( $N_{\text{P}}$  19) feature with low efficiency ( $A_1$  = 52.13) and Pearson correlation ~0.041. Such a combination of features with high and low correlation is observed very often, and this combination results in a high classification efficiency. Among the features from Table 7, the following have a low linear correlation with the diagnosis: MCH (0.041), UA (0.066), amylase (0.03) and LDH (0.071). Pearson's coefficient from the distribution in Figure 3 is indicated in brackets.

There are pairs consisting entirely of effective features, for example, LDL–MCHC ( $N^{\circ}$  43- $N^{\circ}$  20), HDL-C–MCHC ( $N^{\circ}$  36- $N^{\circ}$  20), etc. Figure 6 shows the relationship between the feature pairs for the top 50 results. The main six features are in the center. Asterisks indicate the features that most often form a pair with the main features: UA ( $N^{\circ}$  51), LDH ( $N^{\circ}$  42), CK-MB ( $N^{\circ}$  32) and ALP ( $N^{\circ}$  30). The main feature LDL ( $N^{\circ}$  43) forms the largest number of effective pairs for classification.



**Figure 6.** Pairs of features with high classification efficiency SARS-CoV-2-RBV1 dataset for the Histogram-based Gradient Boosting classifier.

To find the reasons for the effectiveness of the pairs of features from Table 7, twodimensional distributions of the diagnosis (attractors) were constructed for the first six pairs (Figure 7). For the healthy patients (non-COVID-19), there are clear linear and cruciform attractors, while for people diagnosed with COVID-19, these attractors shift and become chaotic. This difference in the shape of the attractors allows for classifiers to effectively distinguish between the two classes. The best separation of attractor shapes is observed for the MCHC–MCH pair (Figure 7a) that explains its highest classification ability. For the pairs in Figure 7b–e, shifted cruciform attractors are observed, which also contributes to their effective separation by classifiers. In Figure 7f, two attractors are blurred, but due to their weak intersection, the classification efficiency is high.



**Figure 7.** Two-dimensional distributions (attractors) of a COVID-19 and non-COVID-19 diagnosis in the coordinates of feature pairs MCHC–MCH (**a**), LDL–CK-MB (**b**), HDL-C–CK-MB (**c**), Triglyceride–CK-MB (**d**), LDL–Cholesterol (**e**), LDL–MCHC (**f**).

When using two features, the maximum accuracy is  $A_2 = 99.81\%$  and this value is lower than when using the 51 features  $A_{51} = 100\%$ . However, feature reduction is important to simplify the classification of patients in practical terms. More accurate models can be obtained using three features.

4.2.3. The Study of the Most Significant Combination of Three Features of the HGB Model

Table 8 presents the classification result of the SARS-CoV-2-RBV1 dataset for the HGB model using three input features.

 

 Table 8. Classification efficiency of SARS-CoV-2-RBV1 dataset using 3 features for the Histogrambased Gradient Boosting classifier.

N≞	First Feature	Second Feature	Third Feature	Average Accuracy $A_3,\%$
39-48-32	Cholesterol	Triglyceride	CK-MB	99.91
39-36-32	Cholesterol	HDL-C	CK-MB	99.91
43-20-19	LDL	MCHC	MCH	99.91
20-31-19	MCHC	Amylase	MCH	99.85
43-51-32	LDL	ŪA	CK-MB	99.85
39-20-19	Cholesterol	MCHC	MCH	99.83
48-42-32	Triglyceride	LDH	CK-MB	99.83
36-20-19	HDL-C	MCHC	MCH	99.79
36-42-32	HDL-C	LDH	CK-MB	99.79
43-38-51	LDL	Cholesterol	UA	99.79
20-48-19	MCHC	Triglyceride	MCH	99.77
39-48-31	Cholesterol	Triglyceride	Amylase	99.77
39-36-38	Cholesterol	HDL-C	Chlorine	99.75
36-31-51	HDL-C	Amylase	UA	99.75
39-36-42	Cholesterol	HDL-C	LDH	99.75
20-51-19	MCHC	UA	MCH	99.74
39-48-38	Cholesterol	Triglyceride	Chlorine	99.72
39-31-51	Cholesterol	Amylase	UA	99.70
39-48-42	Cholesterol	Triglyceride	LDH	99.66
48-31-42	Triglyceride	Amylase	LDH	99.51

An analysis of Table 8 reveals that no new features have been added in the first twenty most accurate models compared to Table 7. The MCH and MCHC features are found only in pairs. With the addition of the third feature, the maximum classification efficiency increased from  $A_2 = 99.81\%$  to  $A_3 = 99.91\%$ .

### 4.2.4. The Study of the Most Significant Combination of 11 Features of the HGB Model

Tables 7 and 8 include only 11 features: LDL ( $\mathbb{N}_{2}$  43), cholesterol ( $\mathbb{N}_{2}$  39), HDL-C ( $\mathbb{N}_{2}$  36), MCHC ( $\mathbb{N}_{2}$  20), triglyceride ( $\mathbb{N}_{2}$  48), amylase ( $\mathbb{N}_{2}$  31), UA ( $\mathbb{N}_{2}$  51), LDH ( $\mathbb{N}_{2}$  42), CK-MB ( $\mathbb{N}_{2}$  32), ALP ( $\mathbb{N}_{2}$  30) and MCH ( $\mathbb{N}_{2}$  19). The classification accuracy of the HGB model using 11 features was  $A_{11} = 100\%$ . Therefore, 11 features are sufficient to determine the presence of COVID-19 using machine learning methods based on the histogram-based gradient boosting classifier.

#### 4.3. LogNNet Implementation on Arduino for Edge Computing

A compact 77-line LogNNet algorithm was created for diagnosing and predicting COVID-19 disease using routine blood values on an Arduino controller.

LogNNet testing on Arduino revealed an accuracy of  $A_{51} = 99.7\%$ , which coincides with the accuracy on the model computer program [46]. The classification time for the input vector is about 0.1 s. An estimate of the RAM used by the Arduino controller is shown in Figure 8.



**Figure 8.** Estimation of the RAM used by the Arduino controller when working with the neural network LogNNet 51:50:20:2.

Input data (array Y)

Global variables (arrays  $S_h$ ,  $S_{h2}$  and variables) occupy 294 bytes of RAM, and incoming data is written to array Y, which occupies 208 bytes. The Arduino uses the Serial system library to operate the serial port. It is loaded at initiation in the "void setup" block and takes 310 bytes of RAM. The data stored in the LogNNet.h library are also loaded into the RAM during the program's initialization and take 2526 bytes, the maximum contribution made by the matrix  $W_1$ —2142 bytes. For local computations within functions and procedures, at least 1012 bytes must be reserved. The total RAM consumption is of 4350 bytes.

# 4.4. Machine Learning COVID-19 Sensor for IoT

The LogNNet network can be easily imported to various microcontrollers and used to predict a diagnosis based on blood biochemical parameters. However, our experimental results in Sections 3.2 and 3.3 are significantly inferior in accuracy to resource-intensive machine learning algorithms. Therefore, we proposed two architectures of the IoT system (Figure 9), which include an IoT device with LogNNet implementation (edge computing) and a cloud service containing a trained HGB model (AI computing). These configurations implement the prognosis of the disease in offline and online modes with ML sensors for diagnosis of the COVID-19 disease (Sensor 1.0 type).



**Figure 9.** Two architectures of the IoT system, which includes an IoT device with LogNNet implementation (edge computing) and a cloud service containing a trained HGB model (AI computing).

In the IoT device, the results of a biochemical blood test are entered manually or transmitted directly from the laboratory equipment. If the cloud service is unavailable or if the blood tests are performed on site using a mobile laboratory in remote areas, the diagnosis is made by the LogNNet network. If the IoT device has access to the network, it sends a network request to the cloud service, wherein the diagnosis is determined using the HGB model. The cloud service sends a response with a diagnosis, which is displayed on the IoT device using an LED indication or on an LCD display.

### 5. Discussion

# 5.1. Analysis of Results from a Medical Perspective

COVID-19 has a higher mortality and infectivity than influenza [3,13]. The disease still causes death and continues to spread [1,6,15]. The use of vaccines did not stop the spread of the disease, and important mutations were detected in the structure of the virus during the epidemic [1]. Most of the infected patients had mild symptoms and a good prognosis. However, some patients developed severe symptoms, such as severe pneumonia, acute respiratory distress syndrome (ARDS) and multiple organ dysfunction syndromes (MODSs) [2,5,24]. A need for studies to determine the prognosis and immune conditions of the COVID-19 disease remains [3,74]. Therefore, the early evaluation of patients who need intensive care and have high mortality expectations as well as the effective identification of relevant biomarkers are important to reduce the mortality of the disease [5,6,25].

Various complications may be encountered during the treatment process of COVID-19 and, therefore, the course of the disease should be predicted earlier [64,77]. It is important to diagnose and predict the prognosis of the disease at an early stage so that the first response to severely infected COVID-19 patients can be conducted properly [2,5].

Although many studies on COVID-19 have been published, the relationships between the pathological aspects of the disease and routine blood values have not been fully determined [77]. Previous studies have reported that changes in many RBVs and hematological abnormalities are observed during the course of the disease [14,77].

In this study, according to the *A*<sub>th</sub> threshold classification result based on [12], the most effective features in the diagnosis of the disease were found to be LDL with 96.47%, HDL-C with 94.73%, cholesterol with 94.47% and MCHC with 94.35% (Table 3). Indeed, in previous studies, large changes in these features were reported in severe and fatal COVID-19 patients, and these features may be important biomarkers for the prognosis of the disease [1,2,5,14].

Considering the linear dependency structures of the features among each other, the most effective combinations of dual features in the diagnosis of the disease were obtained and the Pearson correlation values were calculated (Table 4). The highly positive linear correlation structure of some trait pairs with positive and negative individuals was remarkable. The highest positive and negative linearly correlated trait pairs (High-High) were HCT–HGB, MPV–PDW and HCT–RBC (96%, 93% and 87%, respectively). These features vary greatly in severe COVID-19 patients and may be associated with the prognosis and mortality of the disease [1,2,14]. The high positive association of trait pairs expressed as the High-High type with both positive and negative COVID-19 individuals led us to believe that various comorbidities such as hypertension, obesity and diabetes may exist in our negative COVID-19 population. Considering hospital admissions of negative COVID-19 patients, these trait pairs are highly associated not only with COVID-19 but also with various inflammatory syndromes and infections [1,2,78]. Djakpo et al. [79] stated that the abnormalities of HGB, HCT and RBC or anemia observed in patients with comorbidities are due to the inability of the bone marrow to produce enough RBCs to carry oxygen and lung damage caused by COVID-19 which complicates gas exchange.

Considering the relationship of the patients with these features, the presence of possible comorbid conditions prevents erythrocyte production due to existing inflammation. Since the variation in these trait pairs is hypersensitive to the immune response in individuals, these trait pairs were highly correlated with sick and healthy individuals. The MCH–MCHC trait pair was found to be highly positively correlated, especially with healthy

nt marker to distinguish healthy in

individuals, and this pair may be used as an important marker to distinguish healthy individuals in the diagnosis of the disease. Changes in these characteristics may indicate the suppression of lymphocytic and erythrocyte series or platelet and erythrocyte deformities [1]. In addition, in this study, a highly positive association of the MCH–MCV trait pair with COVID-19 was found. Mertoğlu et al. [2], Huyut et al. [14] and Karakike et al. [21] stated that this was due to the decrease in the size of erythrocytes and anisocytosis in patients. The high positive association of the HGB–RBC trait pair with sick individuals may be related to impaired erythropoiesis in the later stages of the disease. The High–High type feature set provides important clues in the isolation of both sick and healthy individuals.

In Table 4, a high (77%) negative relationship between the fibrinogen–LYM feature pair and COVID-19 patients is seen, and we believe that this level of relationship is due to the fibrinogen feature. Indeed, Winata and Kurniawan [28] noted that the degradation product of fibrinogen (FBU) was increased in all patients in the late stage of COVID-19 and that this was significantly associated with coagulation. In addition, the high correlation of the cholesterol–LDL, cholesterol–HDL-C and chlorine-sodium trait pairs (High–Low type in Table 4) with sick individuals showed that these trait pairs were important markers in identifying sick individuals. Fang et al. [80] and Mertoğlu et al. [1] stated that this feature set may be associated with multi-organ involvement in COVID-19 and the widespread distribution of angiotensin-converting enzyme receptors in the body.

The fact that the Low–High trait pair MCHC–MCV was found to be highly positively correlated with COVID-19 negative individuals in Table 4 suggested the importance of the size of erythrocyte and anisocytosis in healthy individuals [80]. In addition, the functional properties of the ALT–AST, eGFR–Urea and D-Bil–T-Bil pairs were found to be important markers in the isolation of COVID-19 negative individuals. Mertoğlu et al. [1], Huyut et al. [14] and Zhou et al. [27] stated that the decrease in ALT, AST, GGT, total bilirubin and eGFR indicated that the patients had serious damage to organs such as pancreas and kidney. In another study, Bertolini et al. [81] stated that AST, GGT, ALP and bilirubin may be frequently elevated in COVID-19 and that the main underlying causes of this condition may be hyper inflammation and thrombotic microangiopathy. In addition, the high positive correlation of the INR–PT trait pair with negative COVID-19 individuals suggested that it is important to monitor these individuals for the development of disseminated intravascular coagulopathy and acute respiratory distress [80,82].

In this study, 13 popular classifier machine learning models and the LogNNet neural network model were run on 51 routine blood values to detect patients infected with COVID-19. Histogram-based gradient boosting (HGB) was the model with the fastest and highest accuracy in determining the diagnosis of the disease (accuracy: 100%, time: 6.39 sec).

For the HGB model using a single input feature ( $A_1$ ), the most effective features in the diagnosis of the disease were LDL 96.87% ( $\mathbb{N}$  43), cholesterol 9507% ( $\mathbb{N}$  39), HDL-C 94.99% ( $\mathbb{N}$  36), MCHC 94.35% ( $\mathbb{N}$  20), triglyceride ( $\mathbb{N}$  48) and amylase ( $\mathbb{N}$  31) (Table 6). For the HGB model using the dual entry feature ( $A_2$ ), the most effective trait pair in the diagnosis of the disease was MCHC–MCH ( $A_2 = 99.81$ ) (Table 7). The success of MCH as a single-entry feature in the diagnosis of the disease is low ( $A_1 = 52.13$ ). Huyut and Velichko [12] found an accuracy rate of 99.1% in the diagnosis of the disease by running the MCHC–MCH features with LogNNet. The HGB model operated with MCHC–MCH was found to be more successful than the LogNNet model in the diagnosis of the disease.

Since low values of MCH and high values of MCHC were associated with COVID-19 [83], it was expected that the use of these two features together in the diagnosis of the disease would produce higher classification success. The most effective dual trait pairs (Table 7) were similar to the most effective single traits (Table 6) for the HGB model for the diagnosis of the disease. This provides important information about the functional properties of the binary trait pairs obtained with the HGB model in the diagnosis of the disease. Six basic features, that is LDL ( $N^{\circ}$  43), cholesterol ( $N^{\circ}$  39), HDL-C ( $N^{\circ}$  36), MCHC ( $N^{\circ}$  20), triglyceride ( $N^{\circ}$  48) and amylase ( $N^{\circ}$  31), among the combinations of binary features used in the diagnosis of the disease and four features, that is UA ( $N^{\circ}$  51), LDH ( $N^{\circ}$  42), CK-MB ( $N^{\circ}$  32) and ALP ( $N^{\circ}$  30), that most frequently pair with these features are given in Figure 6. The main feature LDL ( $N^{\circ}$  43) generated the largest number of effective pairs for classification. The effectiveness of these feature pairs (Table 7) in detecting patients is visualized in two-dimensional space (Figure 7). Classification is most clearly visible in the MCHC–MCH pair (Figure 7a), which explains the higher classification ability.

In the binary feature combinations used by HGB in the diagnosis of the disease, the maximum accuracy was  $A_2 = 99.81\%$  which is slightly lower than the use of 51 features ( $A_{51} = 100\%$ ). However, feature reduction provides more cost effective and rapid results in interpreting the classification of patients from a practical point of view and identifying the most effective features. The highest classification success obtained for the HGB model using three feature combinations was  $A_3 = 99.91$  (Table 8).

Analysis of Table 8 showed that no new features were added to the top twenty models with the highest accuracy compared to Table 7. The binary combinations in Table 7 were sufficient for the diagnosis of the disease. In addition, the co-existence of MCH and MCHC features in all combinations reveals hidden association structures between these features and contains important clues in the diagnosis of the disease.

In this study, the most important 11 biomarkers were found with the HGB model used to determine the diagnosis of the disease, and with these features, all patients and healthy individuals were correctly identified with high performance (A11 = 100%). In addition, the importance of various combinations of these features in the diagnosis of the disease was recognized. The performance of these 11 features, namely LDL ( $N^{\circ}$  43), cholesterol ( $N^{\circ}$ 39), HDL-C ( $N^{\circ}$  36), MCHC ( $N^{\circ}$  20), triglyceride ( $N^{\circ}$  48), amylase ( $N^{\circ}$  31), UA ( $N^{\circ}$  51), LDH ( $N^{\circ}$  42), CK-MB ( $N^{\circ}$  32), ALP ( $N^{\circ}$  30) and MCH ( $N^{\circ}$  19) and their various combinations in the diagnosis of the disease was higher than the individual performances, suggesting that there is a high level of confidential information between these feature combinations and COVID-19.

Kocar et al. [84] and Zinellu et al. [85] presented evidence of significant changes in the lipid profile of severe COVID-19 patients, particularly in total cholesterol, LDL and HDL-C concentrations. They also reported that increased cholesterol concentrations in the cell membrane increased the binding activity of SARS-CoV-2, facilitated membrane fusion and enabled the successful entry of the virus into the host. Therefore, Kocar et al. [84] and Wei et al. [86] indicated that total cholesterol, LDL and HDL-C characteristics may aid in early risk stratification and clinical decisions. However, conflicting results have been reported for changes in triglyceride levels of severe COVID-19 patients [85]. Stephens et al. [87] stated that in severe COVID-19 patients, the elevated serum amylase value is often not attributable to acute pancreatitis or a clinically significant pancreatic injury, but is more likely to be a nonspecific manifestation of shock/critical illness. Mao et al. [83] stated that changes in leukocytes, neutrophils, lymphocytes, platelets, hemoglobin levels, MCV and MCHC are generally associated with lung involvement, oxygen demand and disease activity. They also noted that high MCV and low MCHC are associated with advanced anemia and are independent predictors of disease worsening [83].

Wu et al. [88] stated that an increase or decrease in LDH is indicative of radiographic progression or improvement. They also demonstrated the potential usefulness of serum LDH as a marker for assessing clinical severity, monitoring treatment response and thus aiding risk stratification and early intervention in COVID-19 pneumonia. Hu et al. [89] stated that SARS-CoV-2 infection is associated with low serum uric acid (SUA) levels, and this feature may be an independent risk factor for the disease. They also noted that male patients with COVID-19 accompanied by low SUA levels are at higher risk of developing severe symptoms than those with high SUA levels at admission. Zinellu et al. [90] found that high CK-MB concentrations were significantly associated with severe morbidity and mortality in COVID-19 patients. They stated that this biomarker of myocardial damage may be useful for the classification of patients with severe COVID-19, and that high CK-MB values may reflect excessive inflammation status. They also stated that the evaluation of CK-MB in COVID-19 patients provides specific clinical information for early risk stratification,

independent of myocardial necrosis and cardiac complications. Afra et al. [91] showed the incidence of abnormal liver tests in severe COVID-19 patients and reported the association of elevated AST, ALT and total bilirubin levels with liver injury in severe COVID-19 patients [13,91,92]. However, conflicting results have been reported regarding the ALP levels of mild and severe COVID-19 patients [91]. In addition, Afra et al. [91] showed that elevated liver enzymes can effectively predict hospital-critical COVID-19 cases.

The accuracy of ML algorithms is difficult to determine when used without any physician input [93,94]. A major limitation of ML is that it is difficult to explain how these algorithms arrive at their conclusions [95]. An ML algorithm can be likened to a black box that takes inputs and produces outputs without any explanation as to how it produces the outputs [94].

Additionally, if an algorithm misdiagnoses a malignant lesion, the algorithm cannot explain why it chose a particular diagnosis [94,95]. While the printouts can aid interpretation, it can be a potential danger and problem to the patient if the model fails to explain to a patient why he or she has diagnosed a lesion as benign or malignant, or how it has chosen a particular treatment [94].

Physician interpretation is necessary for choosing a diagnosis or treatment. In addition to the black box nature of these algorithms, machine learning is also prone to the "garbage in, garbage out" motto [94]. This maxim indicates that the quality of the dataset input determines the quality of the output. Therefore, if the data inputs are badly labeled, the outputs of the algorithm will reflect these inaccuracies [93–95].

In addition, all the devices should be evaluated in prospective clinical trials and made publicly available in the peer-reviewed literature.

#### 5.2. Analysis of Results from IoT Perspective

The aim of this study is the feasibility analysis of a fast, reliable and cost-effective digital tool for the diagnosis of COVID-19 based on the RBV values measured at admission. The proposed solution is based on the concept of ML sensors for diagnosis of the COVID-19 disease (Sensor 1.0 type). The concept makes a step towards "smart sensorics of human" with promising opportunities for AI applications in healthcare.

In our study, we are not targeting IoT systems for telemedicine, wherein any procedure is performed by a physician using telecommunication means of transmitting medical data. Solutions of clinical telemedicine are subject to strict certification. Telemedicine is prescribed by a doctor and is administered via a medical device (never via a smartphone). Instead, we focus on promising IoT systems for telehealth/telecare and mobile health (m-Health) [45]. First, data from ML sensors support the prognosis of the disease in offline and online modes. Second, ML sensors can be used in AAL and other IoT environments to support a person in his/her everyday life. Importantly, COVID-19 is not the only disease to apply ML sensors in IoT systems.

AI methods become effective for the prognosis of various diseases. COVID-19 has opened the new era for AI methods to mitigate future pandemics. The rapidly growing number of publications confirms the potential of ML sensors for collecting datasets for further analysis with AI methods. Predictive analytics uses available retrospective data and various predictive models (including ML-constructed) to aid in answering the question "What could happen?". Prognosis from the sensed data is required not only for clinical medicine (to support clinical medical decisions). Managerial predictive analytics supports managers in healthcare at various levels to assess possible scenarios for the development of diseases, the budgets of medical organizations, the need for medicines, etc.

In AAL, ML sensors are useful in personal use as digital assistance (recommendations, including prognosis). In fact, the five natural human sensors (vision, hearing, touch, smell, taste) are extended by ML sensors. A person can develop health insight from their own RBVs in real time or collect the data for retrospective analysis. Humans themselves can act as complicated sensors [44]. A human traditionally finds a way to enhance her/his function, e.g., glasses (optical tool) to advance the vision or thermometer (physical tool)

to regularly sense body temperature. Now the era of digital tools for personal health assistance is coming.

The implemented prototype of the diagnosis tool demonstrates that the LogNNet network can be imported to various microcontrollers. Many IoT devices can be made smarter, opening a way to develop advanced AmI healthcare with essential parts of IoT and edge intelligence [96]. A LogNNet-equipped ML sensor can be effectively employed in future IoT applications for healthcare and for other problem domains that require active digitalization and emerging AmI methods [46].

The LogNNet network can be used to predict a diagnosis based on blood biochemical parameters. This result is an important step in smart human sensors for IoT application, as the COVID-19 status and other blood-related health parameters are difficult to analyze on the IoT edges (in contrast to more widespread parameters, such as temperature or heartbeat) [97–99]. Our approach is applicable to the development of personalized bionic systems (smart suit for a person or AmI environment with people), wherein disease status recognition is a regular digital service for healthcare or well-being applications in everyday life [45].

Although the small IoT devices cannot provide such high accuracy as resourceintensive ML algorithms on powerful computer systems, AAL systems are intended for everyday life settings (e.g., at home, workspace, outdoor). Where strict medical decisions and critical medical support are not mandatory, the digital services may provide attention points and optional recommendations for personal use. We believe this type of smart human sensors will soon diffuse from the restricted medical lab setting toward the wide market of smart consumer electronics and digital services [100].

#### 6. Limitations of the Study

The data primarily represent a single institution (EBYU-MG) and the Turkish population. Secondly, our dataset does not include comorbidities of patients and other diagnostic information of patient subgroups. In practice, the data in retrospective studies collected in a certain period cannot meet all data sample requirements. We suggest the findings in this study be supported by a retrospective cohort study setup.

## 7. Conclusions and Future Studies

Determining a COVID-19 infected status with diagnostic tests and imaging results is costly and time-consuming. If this process is prolonged, the patient's health may be at greater risk by being exposed to different complications. This study provides a fast, reliable and cost-effective alternative mobile tool for the diagnosis of COVID-19 based on the RBVs measured at the time of admission.

In this study, 13 popular classifier machine learning models and the LogNNet neural network model were run on 51 routine blood values to detect patients infected with COVID-19. The histogram-based gradient boosting (HGB) model was the most successful classification model in terms of accuracy and time in detecting the diagnosis of the disease (accuracy: 100%, time: 6.39 s). In addition, the absence of any normalization method and additional feature selection procedure for the HGB model contributes to the speed and efficiency of the model.

The eleven most important biomarkers in the diagnosis of the disease were found with the HGB classifier: LDL ( $N_{\circ}$  43), cholesterol ( $N_{\circ}$  39), HDL-C ( $N_{\circ}$  36), MCHC ( $N_{\circ}$  20), triglyceride ( $N_{\circ}$  48), amylase ( $N_{\circ}$  31), UA ( $N_{\circ}$  51), LDH ( $N_{\circ}$  42), CK-MB ( $N_{\circ}$  32), ALP ( $N_{\circ}$  30) and MCH ( $N_{\circ}$  19). Using only these 11 RBVs features, the HGB model accurately detected all COVID-19 patients ( $A_{11} = 100\%$ ).

The high accuracy of the single, double and triple combinations of these 11 features selected by the HGB model in the diagnosis of the disease showed the importance of these features in the diagnosis of the disease. In addition, the performance of double and triple combinations of these features in the detection of sick and healthy individuals was higher than the individual performances, suggesting that there is a high level of hidden information between these blood feature combinations and COVID-19.

The HGB model reveals that 11 features are sufficient for the diagnosis of the presence of COVID-19 using the HGB classifier. These features and their binary combinations are an important source of variation in the diagnosis of COVID-19. We propose to use these features and their binary combinations to be run with HGB as important biomarkers in the diagnosis of the disease.

The study results can be effectively used in IoT medical edge devices with low RAM resources, ML sensors, portable point-of-care blood testing devices [101], decision support systems, telecare and m-Health. This opportunity empowers the development of many innovative applications for predictive analytics in clinical MIS or everyday AAL systems.

The artificial intelligence models for the early prediction of the diagnosis and progression of COVID-19 and other diseases produce satisfactory results. Future artificial intelligence studies for the early diagnosis and prognosis of fatal, costly and severe diseases will ease the burden of healthcare professionals and increase patient comfort. In addition, the use the physiological, comorbidity and demographic features of the patients together with the RBVs data may provide interesting insights. Testing the results of this study on multi-racial, multi-center and larger patient groups may improve the generalizability of the findings. In this context, this study may pave the way for many exciting subsequent investigations.

Author Contributions: Conceptualization, M.T.H. and A.V.; methodology, A.V., M.T.H., M.B., Y.I. and D.K.; software, A.V., M.T.H., M.B. and Y.I.; validation, M.T.H. and A.V.; formal analysis, A.V., M.T.H., M.B, Y.I. and D.K.; investigation, A.V.; resources, M.T.H.; data curation, M.T.H.; writing—original draft preparation, A.V., M.T.H., M.B., Y.I. and D.K.; writing—review and editing, A.V., M.T.H., M.B., Y.I. and D.K.; visualization, A.V. and M.B.; supervision, D.K.; project administration, D.K.; funding acquisition, D.K. All authors have read and agreed to the published version of the manuscript.

**Funding:** The research is implemented with financial support by Russian Science Foundation, project no. 22-11-20040 (https://rscf.ru/en/project/22-11-20040/ (accessed on 14 September 2022)) jointly with Republic of Karelia and funding from Venture Investment Fund of Republic of Karelia (VIF RK).

**Institutional Review Board Statement:** The dataset used in this study was collected in order to be used in various studies in the estimation of the diagnosis, prognosis and mortality of COVID-19. The necessary permissions for the collected dataset were given by the Ministry of Health of the Republic of Turkey and the Ethics Committee of Erzincan Binali Yıldırım University. This study was conducted in accordance with the 1989 Declaration of Helsinki. Erzincan Binali Yıldırım University Human Research Health and Sports Sciences Ethics Committee Decision Number: 2021/02-07.

**Informed Consent Statement:** In this study, a dataset including only routine blood values, RT-PCR results (positive or negative) and treatment units of the patients was downloaded retrospectively from the information system of our hospital in a digital environment. A new sample was not taken from the patients. There is no information in the dataset that includes identifying characteristics of individuals. It was stated that routine blood values would only be used in academic studies, and written consent was obtained from the institutions for this. In addition, therefore, written informed consent was not administered for every patient.

**Data Availability Statement:** The data used in this study can be shared with the parties, provided that the article is cited.

Acknowledgments: We thank the method of Erzincan Mengücek Gazi Training and Research Hospital for their support in reaching the material used in this study. Special thanks to the editors of the journal and to the anonymous reviewers for their constructive criticism and improvement suggestions.

Conflicts of Interest: The authors declare no conflict of interest.

# Appendix A

Algorithm A1. LogNNet neural network executable code on Arduino Nano IoT						
1	#include "LogNNet.h"	40				
2		41	byte Output_Layer() {			
3	float Y[S+1];	42	float Sout[N+1]; byte digit = 0;			
4	float Sh[P+1];	43	for (int $j = 0; j \le N; j++$ ) {			
5	float Sh2[M+1];	44	Sout[j] = 0;			
6		45	for (int $i = 0; i \le M; i++$ )			
7	int i = 0;	46	Sout[j] = Sout[j] + Sh2[i] *			
8	String data;	47	((float)W2[i][i]/scale_factor);			
9		48	Sout[i] = Fun activ(Sout[i]):			
10	float Fun_activ(float x) {	49	}			
11	return $1 / (1 + \exp(-1x))$ :	50	for (int i = 0: i <= N: i++) {			
12	}	51	if (Sout[i] > Sout[digit])			
13	1	52	digit = i			
14	void Reservoir(float *Y) {	53	}			
15	$\log W = C$	54	return digit:			
16	Sh[0] = 1	55	}			
17	for (int $i = 1$ ; $i \le P$ ; $i + 1$ )	56	1			
18	Sh[i] = 0	57	void setup() {			
10	for (int i = 0: i <= $S: i \neq 1$ )	58	Serial begin(115200):			
20	W = (D - K * W) % I	59	)			
20	VV = (D - VV) / (D - L) Sh[i] = Sh[i] + ((float)W/ / L) * V[i]:	60	J			
21	$\operatorname{Sit[J]} = \operatorname{Sit[J]} + ((\operatorname{IIOat}) \vee \vee / L) = \operatorname{I}[\operatorname{IJ}],$	61	void loop() (			
22	sh[i] = ((Sh[i] - (float)minS[i-1])/	62	if (Serial.available() > 0) {			
24	scale_factor) /	()				
24	((float)(maxS[j-1]	63	data = Serial.readStringUntil('T');			
25	- minS[j-1])/scale_factor)) - 0.5	64				
26	- (float)meanS[i-1]/(scale_factor*10):	65	if (data != "FN") {			
27	}	66	Y[i] = data to Float()			
28	}	67	i++·			
<u>-</u> 0 29	1	68	}			
30	void Hidden Laver() {	69	) else {			
31	sh2[0] = 1	70	i = 0			
32	for (int i = 1: i <= M: i++) {	71	R = 0, Reservoir(Y):			
33	Sh2[i] = 0	72	Hidden Laver():			
34	for (int i = $0$ ; i <= $P$ ; i = 1)	72	byte Digit - Output I aver();			
35	$Sh2[i] = Sh2[i] \pm Sh[i] *$	74	Serial print(String(Digit)):			
55	5hz[j] – 5hz[j] + 5h[h]	/1	Jenai.philit(Juliig(Digit)),			
36	((float)W1[i][j]/scale_factor);	75	}			
37	Sh2[j] = Fun_activ(Sh2[j]);	76	}			
38	}	77	}			
39	}	78				

# References

- Mertoglu, C.; Huyut, M.; Olmez, H.; Tosun, M.; Kantarci, M.; Coban, T. COVID-19 Is More Dangerous for Older People and Its Severity Is Increasing: A Case-Control Study. *Med. Gas Res.* 2022, 12, 51–54. [CrossRef]
- Mertoglu, C.; Huyut, M.T.; Arslan, Y.; Ceylan, Y.; Coban, T.A. How Do Routine Laboratory Tests Change in Coronavirus Disease 2019? Scand. J. Clin. Lab. Investig. 2021, 81, 24–33. [CrossRef]
- 3. Zhou, C.; Chen, Y.; Ji, Y.; He, X.; Xue, D. Increased Serum Levels of Hepcidin and Ferritin Are Associated with Severity of COVID-19. *Med. Sci. Monit.* 2020, 26, e926178. [CrossRef]
- 4. Huyut, M.T.; Huyut, Z. Forecasting of Oxidant/Antioxidant Levels of COVID-19 Patients by Using Expert Models with Biomarkers Used in the Diagnosis/Prognosis of COVID-19. *Int. Immunopharmacol.* **2021**, *100*, 108127. [CrossRef]
- 5. Huyut, M.T.; İlkbahar, F. The Effectiveness of Blood Routine Parameters and Some Biomarkers as a Potential Diagnostic Tool in the Diagnosis and Prognosis of Covid-19 Disease. *Int. Immunopharmacol.* **2021**, *98*, 107838. [CrossRef]

- 6. Huyut, M.; Üstündağ, H. Prediction of Diagnosis and Prognosis of COVID-19 Disease by Blood Gas Parameters Using Decision Trees Machine Learning Model: A Retrospective Observational Study. *Med. Gas Res.* **2022**, *12*, 60–66. [CrossRef] [PubMed]
- Cabitza, F.; Campagner, A.; Ferrari, D.; Di Resta, C.; Ceriotti, D.; Sabetta, E.; Colombini, A.; De Vecchi, E.; Banfi, G.; Locatelli, M.; et al. Development, Evaluation, and Validation of Machine Learning Models for COVID-19 Detection Based on Routine Blood Tests. *Clin. Chem. Lab. Med.* 2021, 59, 421–431. [CrossRef] [PubMed]
- Vogels, C.B.F.; Brito, A.F.; Wyllie, A.L.; Fauver, J.R.; Ott, I.M.; Kalinich, C.C.; Petrone, M.E.; Casanovas-Massana, A.; Catherine Muenker, M.; Moore, A.J.; et al. Analytical Sensitivity and Efficiency Comparisons of SARS-CoV-2 RT–QPCR Primer–Probe Sets. *Nat. Microbiol.* 2020, 5, 1299–1305. [CrossRef]
- Booth, A.L.; Abels, E.; McCaffrey, P. Development of a Prognostic Model for Mortality in COVID-19 Infection Using Machine Learning. *Mod. Pathol.* 2021, 34, 522–531. [CrossRef]
- Ko, J.; Baldassano, S.N.; Loh, P.-L.; Kording, K.; Litt, B.; Issadore, D. Machine Learning To Detect Signatures of Disease in Liquid Biopsies-A User's Guide Graphical Abstract HHS Public Access. *Lab Chip* 2018, 18, 395–405. [CrossRef]
- Huyut, M.T. Automatic Detection of Severely and Mildly Infected COVID-19 Patients with Supervised Machine Learning Models. IRBM 2022, 1, 35673548. [CrossRef]
- 12. Huyut, M.T.; Velichko, A. Diagnosis and Prognosis of COVID-19 Disease Using Routine Blood Values and LogNNet Neural Network. *Sensors* 2022, 22, 4820. [CrossRef] [PubMed]
- Chen, N.; Zhou, M.; Dong, X.; Qu, J.; Gong, F.; Han, Y.; Qiu, Y.; Wang, J.; Liu, Y.; Wei, Y.; et al. Epidemiological and Clinical Characteristics of 99 Cases of 2019 Novel Coronavirus Pneumonia in Wuhan, China: A Descriptive Study. *Lancet* 2020, 395, 507–513. [CrossRef]
- 14. Tahir Huyut, M.; Huyut, Z.; İlkbahar, F.; Mertoğlu, C. What Is the Impact and Efficacy of Routine Immunological, Biochemical and Hematological Biomarkers as Predictors of COVID-19 Mortality? *Int. Immunopharmacol.* **2022**, *105*, 108542. [CrossRef] [PubMed]
- 15. Huyut, M.T.; Kocaturk, İ. The Effect of Some Symptoms and Features During the Infection Period on the Level of Anxiety and Depression of Adults After Recovery From COVID-19. *Curr. Psychiatry Res. Rev.* **2022**, *18*, 151–163. [CrossRef]
- Guan, W.; Ni, Z.; Hu, Y.; Liang, W.; Ou, C.; He, J.; Liu, L.; Shan, H.; Lei, C.; Hui, D.S.C.; et al. Clinical Characteristics of Coronavirus Disease 2019 in China. N. Engl. J. Med. 2020, 382, 1708–1720. [CrossRef]
- Banerjee, A.; Ray, S.; Vorselaars, B.; Kitson, J.; Mamalakis, M.; Weeks, S.; Baker, M.; Mackenzie, L.S. Use of Machine Learning and Artificial Intelligence to Predict SARS-CoV-2 Infection from Full Blood Counts in a Population. *Int. Immunopharmacol.* 2020, *86*, 106705. [CrossRef] [PubMed]
- 18. Huyut, M.T.; Soygüder, S. The Multi-Relationship Structure between Some Symptoms and Features Seen during the New Coronavirus 19 Infection and the Levels of Anxiety and Depression Post-Covid. *East. J. Med.* **2022**, *27*, 1–10. [CrossRef]
- 19. Perricone, C.; Bartoloni, E.; Bursi, R.; Cafaro, G.; Guidelli, G.M.; Shoenfeld, Y.; Gerli, R. COVID-19 as Part of the Hyperferritinemic Syndromes: The Role of Iron Depletion Therapy. *Immunol. Res.* **2020**, *68*, 213–224. [CrossRef] [PubMed]
- Mehta, P.; McAuley, D.F.; Brown, M.; Sanchez, E.; Tattersall, R.S.; Manson, J.J. COVID-19: Consider Cytokine Storm Syndromes and Immunosuppression. *Lancet* 2020, 395, 1033–1034. [CrossRef]
- Karakike, E.; Giamarellos-Bourboulis, E.J. Macrophage Activation-like Syndrome: A Distinct Entity Leading to Early Death in Sepsis. Front. Immunol. 2019, 10, 55. [CrossRef]
- Tural Onur, S.; Altın, S.; Sokucu, S.N.; Fikri, B.İ.; Barça, T.; Bolat, E.; Toptaş, M. Could Ferritin Level Be an Indicator of COVID-19 Disease Mortality? J. Med. Virol. 2021, 93, 1672–1677. [CrossRef] [PubMed]
- Rosário, C.; Zandman-Goddard, G.; Meyron-Holtz, E.G.; D'Cruz, D.P.; Shoenfeld, Y. The Hyperferritinemic Syndrome: Macrophage Activation Syndrome, Still's Disease, Septic Shock and Catastrophic Antiphospholipid Syndrome. *BMC Med.* 2013, 11, 185. [CrossRef]
- 24. Lippi, G.; Plebani, M.; Henry, B.M. Thrombocytopenia Is Associated with Severe Coronavirus Disease 2019 (COVID-19) Infections: A Meta-Analysis. *Clin. Chim. Acta* 2020, *506*, 145–148. [CrossRef] [PubMed]
- Cheng, L.; Li, H.; Li, L.; Liu, C.; Yan, S.; Chen, H.; Li, Y. Ferritin in the Coronavirus Disease 2019 (COVID-19): A Systematic Review and Meta-Analysis. J. Clin. Lab. Anal. 2020, 34, e23618. [CrossRef] [PubMed]
- Feld, J.; Tremblay, D.; Thibaud, S.; Kessler, A.; Naymagon, L. Ferritin Levels in Patients with COVID-19: A Poor Predictor of Mortality and Hemophagocytic Lymphohistiocytosis. *Int. J. Lab. Hematol.* 2020, 42, 773–779. [CrossRef] [PubMed]
- Zhou, F.; Yu, T.; Du, R.; Fan, G.; Liu, Y.; Liu, Z.; Xiang, J.; Wang, Y.; Song, B.; Gu, X.; et al. Clinical Course and Risk Factors for Mortality of Adult Inpatients with COVID-19 in Wuhan, China: A Retrospective Cohort Study. *Lancet* 2020, 395, 1054–1062. [CrossRef]
- 28. Winata, S.; Kurniawan, A. Coagulopathy in COVID-19: A Systematic Review. Medicinus 2021, 8, 72. [CrossRef]
- Hariyanto, T.I.; Japar, K.V.; Kwenandar, F.; Damay, V.; Siregar, J.I.; Lugito, N.P.H.; Tjiang, M.M.; Kurniawan, A. Inflammatory and hematologic markers as predictors of severe outcomes in COVID-19 infection: A systematic review and meta-analysis. *Am. J. Emerg. Med.* 2021, 41, 110–119. [CrossRef]
- Remeseiro, B.; Bolon-Canedo, V. A Review of Feature Selection Methods in Medical Applications. *Comput. Biol. Med.* 2019, 112, 103375. [CrossRef]
- Guyon, I.; Gunn, S.; Nikravesh, M.; Zadeh, L.A. Feature Extraction: Foundations and Applications; Studies in Fuzziness and Soft Computing; Springer: Berlin/Heidelberg, Germany, 2008; ISBN 9783540354888.

- 32. Jiang, S.Q.; Huang, Q.F.; Xie, W.M.; Lv, C.; Quan, X.Q. The Association between Severe COVID-19 and Low Platelet Count: Evidence from 31 Observational Studies Involving 7613 Participants. *Br. J. Haematol.* **2020**, 190, e29–e33. [CrossRef] [PubMed]
- Zheng, Y.; Zhang, Y.; Chi, H.; Chen, S.; Peng, M.; Luo, L.; Chen, L.; Li, J.; Shen, B.; Wang, D. The Hemocyte Counts as a Potential Biomarker for Predicting Disease Progression in COVID-19: A Retrospective Study. *Clin. Chem. Lab. Med.* 2020, 58, 1106–1115. [CrossRef] [PubMed]
- Hall, M.A. Correlation-Based Feature Selection for Machine Learning. Ph.D. Thesis, The University of Waikato, Hamilton, New Zealand, 1999.
- 35. Dash, M.; Liu, H. Consistency-Based Search in Feature Selection. Artif. Intell. 2003, 151, 155–176. [CrossRef]
- Zhao, Z.; Liu, H. Searching for Interacting Features. In Proceedings of the 20th International Joint Conference on Artificial Intelligence, Hyderabad, India, 6–12 January 2007; pp. 1156–1161.
- Hall, M.A.; Smith, L.A. Practical Feature Subset Selection for Machine Learning. In Proceedings of the the 21st Australasian Computer Science Conference ACSC'98, Perth, Australia, 4–6 February 1998; Volume 20, pp. 181–191.
- Kononenko, I. Estimating Attributes: Analysis and Extensions of RELIEF. Machine Learning: ECML-94; Lecture Notes in Computer Science; Springer: Berlin/Heidelberg, Germany, 1994; Volume 784, pp. 171–182. [CrossRef]
- Le Thi, H.A.; Nguyen, V.V.; Ouchani, S. Gene Selection for Cancer Classification Using DCA. Advanced Data Mining and Applications; Lecture Notes in Computer Science; Springer: Berlin/Heidelberg, Germany, 2008; Volume 5139, pp. 62–72. [CrossRef]
- 40. Tibshirani, R. Regression Shrinkage and Selection via the Lasso. J. R. Stat. Society. Ser. B Methodol. 1996, 58, 267–288. [CrossRef]
- Zhu, J.S.; Ge, P.; Jiang, C.; Zhang, Y.; Li, X.; Zhao, Z.; Zhang, L.; Duong, T.Q. Deep-Learning Artificial Intelligence Analysis of Clinical Variables Predicts Mortality in COVID-19 Patients. J. Am. Coll. Emerg. Physicians Open 2020, 1, 1364–1373. [CrossRef] [PubMed]
- Meigal, A.Y.; Korzun, D.G.; Moschevikin, A.P.; Reginya, S.; Gerasimova-Meigal, L.I. Ambient Assisted Living At-Home Laboratory for Motor Status Diagnostics in Parkinson's Disease Patients and Aged People. In *Research Anthology on Supporting Healthy Aging in a Digital Society*; IGI Global: Hershey, PA, USA, 2022; pp. 836–862. Available online: https://services.igi-global.com/ resolvedoi/resolve.aspx?doi=10.4018/978-1-6684-5295-0.ch047 (accessed on 1 October 2022). [CrossRef]
- 43. Fernandes, J.M.; Silva, J.S.; Rodrigues, A.; Boavida, F. A Survey of Approaches to Unobtrusive Sensing of Humans. *ACM Comput. Surv.* **2022**, *55*, 1–28. [CrossRef]
- Kostakos, V.; Rogstadius, J.; Ferreira, D.; Hosio, S.; Goncalves, J. Participatory Sensing, Opinions and Collective Awareness, 1st ed.; Loreto, V., Haklay, M., Hotho, A., Servedio, V.D.P., Stumme, G., Theunis, J., Tria, F., Eds.; Springer International Publishing: Cham, Switzerland, 2017; pp. 69–92. ISBN 978-3-319-25658-0.
- 45. Korzun, D.G. Internet of Things Meets Mobile Health Systems in Smart Spaces: An Overview. In *Internet of Things and Big Data Technologies for Next Generation Healthcare;* Bhatt, C., Dey, N., Ashour, A.S., Eds.; Springer International Publishing: Cham, Switzerland, 2017; pp. 111–129. ISBN 978-3-319-49736-5.
- Korzun, D.; Balandina, E.; Kashevnik, A.; Balandin, S.; Viola, F. Ambient Intelligence Services in IoT Environments: Emerging Research and Opportunities; IGI Global: Hershey, PA, USA, 2019; ISBN 9781522589730.
- 47. Warden, P.; Stewart, M.; Plancher, B.; Banbury, C.; Prakash, S.; Chen, E.; Asgar, Z.; Katti, S.; Reddi, V.J. Machine Learning Sensors. *arXiv* 2022. [CrossRef]
- Chinchole, S.; Patel, S. Artificial Intelligence and Sensors Based Assistive System for the Visually Impaired People. In Proceedings of the 2017 International Conference on Intelligent Sustainable Systems (ICISS), Palladam, India, 7–8 December 2017; pp. 16–19.
- Gulzar Ahmad, S.; Iqbal, T.; Javaid, A.; Ullah Munir, E.; Kirn, N.; Ullah Jan, S.; Ramzan, N. Sensing and Artificial Intelligent Maternal-Infant Health Care Systems: A Review. Sensors 2022, 22, 4362. [CrossRef]
- 50. Machine Learning Sensors: Truly Data-Centric AI | Towards Data Science. Available online: https://towardsdatascience.com/ machine-learning-sensors-truly-data-centric-ai-8f6b9904633a (accessed on 23 August 2022).
- Ma, Y.; Hou, L.; Yang, X.; Huang, Z.; Yang, X.; Zhao, N.; He, M.; Shi, Y.; Kang, Y.; Yue, J.; et al. The Association between Frailty and Severe Disease among COVID-19 Patients Aged over 60 Years in China: A Prospective Cohort Study. *BMC Med.* 2020, 18, 274. [CrossRef]
- 52. Fan, B.E. Hematologic Parameters in Patients with COVID-19 Infection: A Reply. Am. J. Hematol. 2020, 95, E215. [CrossRef]
- 53. Ferrari, D.; Motta, A.; Strollo, M.; Banfi, G.; Locatelli, M. Routine Blood Tests as a Potential Diagnostic Tool for COVID-19. *Clin. Chem. Lab. Med.* **2020**, *58*, 1095–1099. [CrossRef] [PubMed]
- 54. Wu, J.; Zhang, P.; Zhang, L.; Meng, W.; Li, J.; Tong, C.; Li, Y.; Cai, J.; Yang, Z.; Zhu, J.; et al. Rapid and Accurate Identification of COVID-19 Infection through Machine Learning Based on Clinical Available Blood Test Results. *medRxiv* 2020. [CrossRef]
- 55. Shakeel, P.M.; Burhanuddin, M.A.; Desa, M.I. Lung Cancer Detection from CT Image Using Improved Profuse Clustering and Deep Learning Instantaneously Trained Neural Networks. *Meas. J. Int. Meas. Confed.* **2019**, *145*, 702–712. [CrossRef]
- Cui, H.; Wang, X.; Feng, D. Automated Localization and Segmentation of Lung Tumor from PET-CT Thorax Volumes Based on Image Feature Analysis. In Proceedings of the 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society, San Diego, CA, USA, 28 August–1 September 2012; pp. 5384–5387. [CrossRef]
- 57. Tomita, K.; Nagao, R.; Touge, H.; Ikeuchi, T.; Sano, H.; Yamasaki, A.; Tohda, Y. Deep Learning Facilitates the Diagnosis of Adult Asthma. *Allergol. Int.* **2019**, *68*, 456–461. [CrossRef]
- 58. Ryu, K.S.; Lee, S.W.; Batbaatar, E.; Lee, J.W.; Choi, K.S.; Cha, H.S. A Deep Learning Model for Estimation of Patients with Undiagnosed Diabetes. *Appl. Sci.* 2020, *10*, 421. [CrossRef]

- Kolachalama, V.B.; Singh, P.; Lin, C.Q.; Mun, D.; Belghasem, M.E.; Henderson, J.M.; Francis, J.M.; Salant, D.J.; Chitalia, V.C. Association of Pathological Fibrosis With Renal Survival Using Deep Neural Networks. *Kidney Int. Rep.* 2018, *3*, 464–475. [CrossRef] [PubMed]
- 60. Wu, C.C.; Yeh, W.C.; Hsu, W.D.; Islam, M.M.; Nguyen, P.A.A.; Poly, T.N.; Wang, Y.C.; Yang, H.C.; Li, Y.C.J. Prediction of Fatty Liver Disease Using Machine Learning Algorithms. *Comput. Methods Programs Biomed.* **2019**, 170, 23–29. [CrossRef] [PubMed]
- 61. Oguntimilehin, A.; Adetunmbi, A.O.; Abiola, O.B. A Machine Learning Approach to Clinical Diagnosis of Typhoid Fever. *Int. J. Comput. Inf. Technol.* **2013**, *4*, 671–676.
- 62. Kouchaki, S.; Yang, Y.Y.; Walker, T.M.; Walker, A.S.; Wilson, D.J.; Peto, T.E.A.; Crook, D.W.; Clifton, D.A.; Hoosdally, S.J.; Gibertoni Cruz, A.L.; et al. Application of Machine Learning Techniques to Tuberculosis Drug Resistance Analysis. *Bioinformatics* **2019**, *35*, 2276–2282. [CrossRef]
- 63. Taylor, R.A.; Moore, C.L.; Cheung, K.H.; Brandt, C. Predicting Urinary Tract Infections in the Emergency Department with Machine Learning. *PLoS ONE* **2018**, *13*, e0194085. [CrossRef]
- Yang, H.S.; Hou, Y.; Vasovic, L.V.; Steel, P.A.D.; Chadburn, A.; Racine-Brzostek, S.E.; Velu, P.; Cushing, M.M.; Loda, M.; Kaushal, R.; et al. Routine Laboratory Blood Tests Predict SARS-CoV-2 Infection Using Machine Learning. *Clin. Chem.* 2020, *66*, 1396–1404. [CrossRef] [PubMed]
- 65. Brinati, D.; Campagner, A.; Ferrari, D.; Locatelli, M.; Banfi, G.; Cabitza, F. Detection of COVID-19 Infection from Routine Blood Exams with Machine Learning: A Feasibility Study. *J. Med. Syst.* **2020**, *44*, 135. [CrossRef]
- Zhang, S.; Huang, S.; Liu, J.; Dong, X.; Meng, M.; Chen, L.; Wen, Z.; Zhang, L.; Chen, Y.; Du, H.; et al. Identification and Validation of Prognostic Factors in Patients with COVID-19: A Retrospective Study Based on Artificial Intelligence Algorithms. *J. Intensive Med.* 2021, 1, 103–109. [CrossRef]
- 67. Alle, S.; Karthikeyan, A.; Kanakan, A.; Siddiqui, S.; Garg, A.; Mehta, P.; Mishra, N.; Chattopadhyay, P.; Devi, P.; Waghdhare, S.; et al. COVID-19 Risk Stratification and Mortality Prediction in Hospitalized Indian Patients: Harnessing Clinical Data for Public Health Benefits. *PLoS ONE* **2022**, *17*, e0264785. [CrossRef] [PubMed]
- 68. Gao, Y.; Chen, L.; Chi, J.; Zeng, S.; Feng, X.; Li, H.; Liu, D.; Feng, X.; Wang, S.; Wang, Y.; et al. Development and Validation of an Online Model to Predict Critical COVID-19 with Immune-Inflammatory Parameters. *J. Intensive Care* **2021**, *9*, 19. [CrossRef]
- 69. Vaishnav, P.K.; Sharma, S.; Sharma, P. Analytical Review Analysis for Screening COVID-19 Disease. *Int. J. Mod. Res.* 2021, 1, 22–29.
- 70. Kukar, M.; Gunčar, G.; Vovko, T.; Podnar, S.; Černelč, P.; Brvar, M.; Zalaznik, M.; Notar, M.; Moškon, S.; Notar, M. COVID-19 Diagnosis by Routine Blood Tests Using Machine Learning. *Sci. Rep.* **2021**, *11*, 10738. [CrossRef] [PubMed]
- 71. Mei, X.; Lee, H.C.; Diao, K.Y.; Huang, M.; Lin, B.; Liu, C.; Xie, Z.; Ma, Y.; Robson, P.M.; Chung, M.; et al. Artificial Intelligence– Enabled Rapid Diagnosis of Patients with COVID-19. *Nat. Med.* **2020**, *26*, 1224–1228. [CrossRef] [PubMed]
- 72. Gómez-Pastora, J.; Weigand, M.; Kim, J.; Wu, X.; Strayer, J.; Palmer, A.F.; Zborowski, M.; Yazer, M.; Chalmers, J.J. Hyperferritinemia in Critically Ill COVID-19 Patients—Is Ferritin the Product of Inflammation or a Pathogenic Mediator? *Clin. Chim. Acta* 2020, 509, 249–251. [CrossRef]
- 73. Boslaugh, S. Statistics in a Nutshell, 2nd ed.; O'Reilly Media, Incorporated: Newton, MA, USA, 2012; ISBN 9781449361129.
- 74. Weiss, G.; Ganz, T.; Goodnough, L.T. Anemia of Inflammation. *Blood* 2019, 133, 40–50. [CrossRef]
- 75. Harrington, P. Machine Learning in Action; Simon and Schuster: New York, NY, USA, 2012; ISBN 978-1617290183.
- Feurer, M.; Eggensperger, K.; Falkner, S.; Lindauer, M.; Hutter, F. Auto-Sklearn 2.0: Hands-Free AutoML via Meta-Learning. *arXiv* 2020, arXiv:2007.04074.
- Kim, S.; Kim, D.-M.; Lee, B. Insufficient Sensitivity of RNA Dependent RNA Polymerase Gene of SARS-CoV-2 Viral Genome as Confirmatory Test Using Korean COVID-19 Cases. *Preprints* 2020, 1–14. [CrossRef]
- Teymouri, M.; Mollazadeh, S.; Mortazavi, H.; Naderi Ghale-noie, Z.; Keyvani, V.; Aghababaei, F.; Hamblin, M.R.; Abbaszadeh-Goudarzi, G.; Pourghadamyari, H.; Hashemian, S.M.R.; et al. Recent Advances and Challenges of RT-PCR Tests for the Diagnosis of COVID-19. *Pathol. Res. Pract.* 2021, 221, 153443. [CrossRef]
- 79. Djakpo, D.K.; Wang, Z.; Zhang, R.; Chen, X.; Chen, P.; Ketisha Antoine, M.M.L. Blood Routine Test in Mild and Common 2019 Coronavirus (COVID-19) Patients. *Biosci. Rep.* 2020, *40*, BSR20200817. [CrossRef]
- 80. Fang, B.; Meng, Q.H. The Laboratory's Role in Combating COVID-19. Crit. Rev. Clin. Lab. Sci. 2020, 57, 400–414. [CrossRef]
- Bertolini, A.; van de Peppel, I.P.; Bodewes, F.A.J.A.; Moshage, H.; Fantin, A.; Farinati, F.; Fiorotto, R.; Jonker, J.W.; Strazzabosco, M.; Verkade, H.J.; et al. Abnormal Liver Function Tests in Patients With COVID-19: Relevance and Potential Pathogenesis. *Hepatology* 2020, 72, 1864–1872. [CrossRef]
- Terpos, E.; Ntanasis-Stathopoulos, I.; Elalamy, I.; Kastritis, E.; Sergentanis, T.N.; Politou, M.; Psaltopoulou, T.; Gerotziafas, G.; Dimopoulos, M.A. Hematological Findings and Complications of COVID-19. *Am. J. Hematol.* 2020, *95*, 834–847. [CrossRef] [PubMed]
- Mao, J.; Dai, R.; Du, R.C.; Zhu, Y.; Shui, L.P.; Luo, X.H. Hematologic Changes Predict Clinical Outcome in Recovered Patients with COVID-19. Ann. Hematol. 2021, 100, 675–689. [CrossRef]
- Kočar, E.; Režen, T.; Rozman, D. Cholesterol, Lipoproteins, and COVID-19: Basic Concepts and Clinical Applications. *Biochim. Biophys. Acta Mol. Cell Biol. Lipids* 2021, 1866, 209–214. [CrossRef]

- Zinellu, A.; Paliogiannis, P.; Fois, A.G.; Solidoro, P.; Carru, C.; Mangoni, A.A. Cholesterol and Triglyceride Concentrations, COVID-19 Severity, and Mortality: A Systematic Review and Meta-Analysis With Meta-Regression. *Front. Public Health* 2021, 9, 705916. [CrossRef]
- Wei, X.; Zeng, W.; Su, J.; Wan, H.; Yu, X.; Cao, X.; Tan, W.; Wang, H. Hypolipidemia Is Associated with the Severity of COVID-19. J. Clin. Lipidol. 2020, 14, 297–304. [CrossRef] [PubMed]
- 87. Stephens, J.R.; Wong, J.L.C.; Broomhead, R.; Mpfle, R.S.; Waheed, U.; Patel, P.; Brett, S.J.; Soni, S. Raised Serum Amylase in Patients with COVID-19 May Not Be Associated with Pancreatitis. *Br. J. Surg.* **2021**, *108*, e152–e153. [CrossRef]
- Wu, M.Y.; Yao, L.; Wang, Y.; Zhu, X.Y.; Wang, X.F.; Tang, P.J.; Chen, C. Clinical Evaluation of Potential Usefulness of Serum Lactate Dehydrogenase (LDH) in 2019 Novel Coronavirus (COVID-19) Pneumonia. *Respir. Res.* 2020, 21, 171. [CrossRef] [PubMed]
- 89. Hu, F.; Guo, Y.; Lin, J.; Zeng, Y.; Wang, J.; Li, M.; Cong, L. Association of Serum Uric Acid Levels with COVID-19 Severity. *BMC Endocr. Disord.* 2021, 21, 97. [CrossRef] [PubMed]
- 90. Zinellu, A.; Sotgia, S.; Fois, A.G.; Mangoni, A.A. Serum CK-MB, COVID-19 Severity and Mortality: An Updated Systematic Review and Meta-Analysis with Meta-Regression. *Adv. Med. Sci.* 2021, *66*, 304–314. [CrossRef]
- Afra, H.S.; Amiri-Dashatan, N.; Ghorbani, F.; Maleki, I. Positive Association between Severity of COVID-19 Infection and Liver Damage: A Systematic Review and Meta-Analysis. *Gastroenterol. Hepatol. Bed Bench* 2020, 13, 292–304. [CrossRef]
- 92. Huang, Y.; Yang, R.; Xu, Y.; Gong, P. Clinical Characteristics of 36 Non-Survivors with COVID-19 in Wuhan, China. *medRxiv* 2020. [CrossRef]
- González-Cruz, C.; Jofre, M.A.; Podlipnik, S.; Combalia, M.; Gareau, D.; Gamboa, M.; Vallone, M.G.; Faride Barragán-Estudillo, Z.; Tamez-Peña, A.L.; Montoya, J.; et al. Machine Learning in Melanoma Diagnosis. Limitations About to Be Overcome. *Actas Dermo-Sifiliográficas* (*Engl. Ed.*) 2020, 111, 313–316. [CrossRef] [PubMed]
- 94. Chan, S.; Reddy, V.; Myers, B.; Thibodeaux, Q.; Brownstone, N.; Liao, W. Machine Learning in Dermatology: Current Applications, Opportunities, and Limitations. *Dermatol. Ther.* **2020**, *10*, 365–386. [CrossRef] [PubMed]
- 95. Sáez, C.; Romero, N.; Conejero, J.A.; García-Gómez, J.M. Potential Limitations in COVID-19 Machine Learning Due to Data Source Variability: A Case Study in the NCov2019 Dataset. *J. Am. Med. Inform. Assoc.* **2021**, *28*, 360–364. [CrossRef]
- Amin, S.U.; Hossain, M.S. Edge Intelligence and Internet of Things in Healthcare: A Survey. *IEEE Access* 2021, *9*, 45–59. [CrossRef]
   Sruthi, P.L.; Raju, K.B. Prediction of the COVID-19 Pandemic with Machine Learning Models. In Proceedings of the 2021 Fifth International Conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud) (I-SMAC), Palladam, India, 11–13 November 2021; pp. 474–481.
- 98. Anjum, N.; Alibakhshikenari, M.; Rashid, J.; Jabeen, F.; Asif, A.; Mohamed, E.M.; Falcone, F. IoT-Based COVID-19 Diagnosing and Monitoring Systems: A Survey. *IEEE Access* 2022, *10*, 87168–87181. [CrossRef]
- Mukati, N.; Namdev, N.; Dilip, R.; Hemalatha, N.; Dhiman, V.; Sahu, B. Healthcare Assistance to COVID-19 Patient Using Internet of Things (IoT) Enabled Technologies. *Mater. Today Proc.* 2021. [CrossRef]
- 100. Meigal, A.Y.; Korzun, D.G.; Gerasimova-Meigal, L.I.; Borodin, A.V.; Zavyalova, Y.V. Ambient Intelligence At-Home Laboratory for Human Everyday Life. Int. J. Embed. Real-Time Commun. Syst. 2019, 10, 117–134. Available online: https://services.igi-global. com/resolvedoi/resolve.aspx?doi=10.4018/IJERTCS.2019040108 (accessed on 14 September 2022). [CrossRef]
- 101. Vetter, B.; Sampath, R.; Carmona, S. Landscape of Point-of-Care Devices for Testing of Cardiometabolic Diseases; FIND: Geneva, Switzerland, 2020.